# PeterConnects

# **Receptionist User Manual**

# v1.16

Oct 2024





# Contents

1	Note	ote to the reader	6	
2	Sign	gn in page	7	
	2.1	Microsoft sign in	7	
	2.2	Webex sign in	8	
	2.3	Receptionist sign in		
	2.4	Change Receptionist sign in	11	
3	The	e Receptionist main screen		
	3.1	zing the panels1		
	3.2	Full screen mode	13	
4	The	e contact list	15	
	4.1	View modes in the contact list	15	
	4.1.2	L.1 Default mode: Recently used contacts	15	
	4.1.2	I.2 Favorites mode	15	
	4.1.3	1.3 Search mode	16	
	4.1.4	1.4 Title bar text depending on the mode	17	
	4.2	Viewing the details of a contact		
	4.3	Multiple columns	19	
	4.4	Contact types	19	
	4.5	Queues	20	
	4.6	Presence	20	
	4.7	Hook states	21	
	4.8	Actions in the contact list	23	
	4.9	Managing your favorites	24	
	4.9.2	9.1         Adding a favorite	24	
	4.9.2	9.2 Removing a favorite	25	
	4.9.3	0.3 Displaying favorites	25	
	4.9.4	9.4 Sorting favorites	25	
	4.10	Searching and filtering	28	
	4.10	IO.1 Searching a contact	28	
	4.10	IO.2 Filtering contacts from the filter menu		
	4.10	10.3 Saving search queries		
	4.10	10.4 Advanced search		
	4.10	10.5 Copy contact information		

2	4.:	11	Editi	ing contacts	42
		4.11	.1	Adding a contact	42
		4.11	.2	Editing a contact	43
		4.11	.3	Deleting a contact	44
	4.1	12	Shar	red contacts	45
		4.12	.1	Filtering shared contacts	45
		4.12	.2	Adding a shared contact	46
		4.12	.3	Editing a Shared contact	47
		4.12	.4	Deleting a contact	48
	4.1	13	Cont	tact Sources	49
		4.13	.1	Filtering contact sources	50
	4.1	14	Send	ding an email to a contact	50
		4.14.1		Opening the email form in a new window	52
	4.1	4.15 Mal		ing a contact note	53
5		Call	funct	tionality	57
	5.1 Cor		Con	nection with the server	57
	5.2 Call		Call	states, destination and call actions	57
	5.2.1		1	Call states	58
	5.2.2		2	Destination information	58
	5.2.3		3	Call actions	59
		5.2.4	1	Greyed out pickup button	59
	5.3	3	Dire	cted call pickup	60
	5.4	4	Tran	nsferring a call	60
	5.5	5	Call	recording	62
	5.6	5	Call	recording options	63
	5.7	7	Con	ference calls	64
	5.8	8	Wor	king with queue calls	66
		5.8.1		Toggling visibility of all joined or all monitored queues	67
		5.8.2	2	Toggling visibility of a single queue	67
		5.8.3	3	Restriction in picking up queue calls	68
	5.9	9	Calli	ng or transferring to other numbers of a contact	68
		5.9.1	1	Calling to other numbers of a contact	68
		5.9.2	2	Transferring to other numbers of a contact	69
	5.2	10	Sele	cting a device for calls	70

6	6 Receptionist status		eptio	nist status	71
	6.1		Pres	ence	71
	6.2	6.2 Do i		not disturb	71
	6.3	6.3 For		vard all calls	71
7		Pers	onali	zing the Receptionist	73
	7.1	L	Inte	rface preferences	73
	7.2	2	Con	versation settings	75
		7.2.3	1	Call handling	76
		7.2.2	2	Busy camp on	76
		7.2.3	3	Audio Alerting	76
		7.2.4		Calling timeout	77
	7.3	3	Que	ue management	78
	ł	7.3.1		Joined queues	78
		7.3.2		Queues to monitor	80
	7.4	ļ	Avai	lability	81
	7.4.		1	Do not disturb	81
		7.4.2		Forward all calls	81
8		Keyl	board	navigation	85
9	,	Wid	gets.		87
	9.1	9.1 Max		imizing and restoring	87
	9.2	9.2 Mo		ring the widgets around	88
	9.3	9.3 Ad		ing a widget	88
	9.4	ļ	URL	Placeholders for widgets	90
	9.5	5	Ove	rview of all widgets	91
		9.5.:	1	Calendar widget	91
		9.5.2	2	Call flow widget	92
		9.5.3	3	Contact information widget	92
		9.5.4	4	Contact note widget	93
		9.5.	5	Homepage widget	93
		9.5.0	6	My call history widget	93
		9.5.7		Twitter widget	93
		9.5.8	8	Web Page widget	93
10	0	Н	elp fı	unction	95
	10	.1	Rece	eptionist version	95

10	).2 Keyl		poard shortcuts	95
10	.3	User	Manual	95
10	.4	Appl	lication messages	96
10	.5	Wha	t's new	96
11	Si	gn ou	ıt	97
12	Gl	lobal	configuration	98
12.	.1	Cale	ndar	98
	12.1	.1	Microsoft Office 365	99
	12.1	.2	Google Suite	01
12.	.2	Mail		01
	12.2	.1	SMTP Authentication1	02
	12.2	.2	Microsoft Office 365 Authentication1	04
	12.2	.3	Without Authentication1	05
12.	.3	User	privileges via Admin portal1	07
	12.3	.1	Setting user privileges for default user1	08
	12.3	.2	Setting user privileges for specific users	08
	12.3	.3	Manage Profile Feature: User Manual1	10
	12.3	.4		13
12.	.4	Mair	ntain callback mail template1	13
	12.4	.1	Adding a new mail template1	13
	12.4	.2	Arranging the order of mail templates1	14
	12.4	.3	Changing or deleting a mail template or creating a new one1	15
12.	.5	Cont	act sources1	15
	12.5	.1	Managing access1	17
	12.5	.2	Adding or changing a contact source1	19
	12.5	.3	Deleting a contact source1	25
	12.5	.4	Supported file formats and columns1	26
12.	.6	Ente	rprise Group selection1	29
13	Cł	nange	e history1	31

# 1 Note to the reader

This user manual is aimed at users of the Receptionist. The images and screenshots in this document are made to illustrate the described functionality. Because of this, some details might be intentionally left out. Depending on the version, available features and configuration, the actual application might differ from the screenshots.

Vodafone One Net does not support the following functionality in the below chapters.

**Chapter 2.1** Microsoft sign in, this is not applicable because Vodafone One Net does not support Peter Connects with Microsoft Teams plug in.

**Chapter 12** Global Configuration is not supported by Vodafone One Net for Peter Connects users. The global configuration part is responsibility of your own IT configuration for email & agenda.

### 2 Sign in page

When you use the browser to navigate to the receptionist (the URL will be provided by your administrator), you will be redirected to a sign in page.

There are three possible sign in flows, depending on your version of the receptionist:

- 1. Version with Microsoft sign in
- 2. Version with Webex sign in
- 3. Version with Receptionist sign in

#### 2.1 Microsoft sign in

The standard Microsoft sign in window will be displayed.

First enter your Microsoft credentials: username

Sign in	
Email, phone, or Skype	
No account? Create one!	
Can't access your account?	
	Next

Next step is entering the password:

Microsoft	
← stephen.postma@pcadev.ne	t
Enter password	
Password	
Forgot my password	
	Sign in

Last Microsoft sign in step is this, select whatever you feel is appropriate.

Microsoft					
stephen.postma@pcadev.net					
Stay signed in?					
Do this to reduce the number of times you are asked to sign in.					
Don't show this again					
	No	<u>Yes</u>			

Now you are correctly signed in with your Microsoft account.

The second sign in you need to do (only one time) is described in section 2.3: Receptionist sign in.

On subsequent logins, you will not be asked for this second set of credentials anymore, only for your Microsoft credentials.

After providing the correct credentials you will be signed in into the Receptionist.

#### 2.2 Webex sign in

The standard Webex sign in window will be displayed.

First enter your Webex credentials: Email address

Welcon	ne to Webex
Email address	
Email address	3

#### Next step is entering the password:



Now you are correctly signed in with your Webex account.

The second sign in you need to do (only one time) is described in section 2.3: Receptionist login.

On subsequent sign in, you will not be asked for this second set of credentials anymore, only for your Microsoft credentials.

After providing the correct credentials you will be signed in into the Receptionist.

#### 2.3 Receptionist sign in

Here you can enter your Receptionist credentials and click 'Sign in'.

Si	gn in
PeterConnect Recep	ets otionist
test@test.com	5
Password	Ø
Si	gn in
Forgot password?	

If you have forgotten your password, you can click on the link 'Forgot password?' below the sign in form. You will be redirected to a webpage where you can reset your password. Note that this link is only visible when your provider has made this feature available.

If you want your username to be remembered the next time you sign in, you can check 'Remember username'.

A specific error code is shown when the sign in fails. An example of such an error is:

test@test.com	<b></b>				
Password	Ø				
PCAUTH0301: Invalid use	rname/password combination				
Sign in					
Forgot password?	Remember username				

This error code can be communicated to the administrator for support.

#### 2.4 Change Receptionist sign in

It is possible to change the connected Receptionist sign in. This is useful when the Microsoft or Webex sign in needs to be connected to another Receptionist account (calling user).

For this use the option 'Sign out calling user' in the Menu.



Next time when you sign in with your Microsoft or Webex credentials, the sign in screen for the Receptionist sign in is shown again.

# 3 The Receptionist main screen

Once you have signed in, you will see the Receptionist screen. The figure below explains the different areas:

PeterConnects	User information and the menu 🕼 monica (64111) 👻 💿 Available	× 🗖 🖗
Conversations (F2)	Homepage	ß
Patrick Troughton (88039) Relipcol.ogistics - Servicedesk © Distinction: Sales important	Startup     Detail area, starting     with the homepage	Q
Conversation area		
Q Search by number, name or annall address		
Allison Price		
Arman Jamesson Contact list	and the second s	
Bethold Prestwood	PeterConnects	
Stathlin Schulz Schulz Selection Cognition	Receptionist	
Chaitan Brahney	1.00000101101	

When you click on a conversation in F2, the details of that conversation will be shown in the F3 screen (Conversation details). Clicking on the 'X' will close this screen.

When you click on a contact in the contact list (F6), the details of that contact will be shown in the F7 screen (Contact details). Clicking on the 'X' will close this screen.

Pete	rConnects			💄 operator65@bw.dev.c	npeterconnects.com (0850068772)	🔴 Busy 🗖 🖬	•
Conve	ersations (F2)	@ ♠   &	Conversation details (	F3)		ĭa(★)	ľ
So My c	alis (0) John van der 65		John 51 (51101)	s ringing 💼 Conversat	ion details		1
J.	(65121) \$ 420:20 - On hold • Destination Operator 55		& Contactinformatie			ଷ୍	
	•	_	First name		John		L
2	John 51 (51101)		Last name		51		L
	Ne Train, Contraportegi tana .	Acres (	Primary phone number		51101		L
Pirst     Prim	ary Queue 53 (0)		Email		stephen.postma@fest.test		l
	=						2
Conto	icts (F6)	2+ 7 i	Contact details (F7)			ĭ ×	
Q	rch by nomber, name, email addres	ss or department	George 55 (+318	50068779) Contact de	etails	P 🗶 🗷 🕓	
05	Operator 54		& Contactinformatie	Q	Contact notes	Q	
0	George 55		First name	George	No notes		
3	+ 0 + r <sub>200</sub> *		Last name	55			
0	John 55		Primary phone number	+31850068779			
-	10 t		Email	stephen.postma2@peterconnects.co	stephen.postma2/@peterconnects.co		
	Monica 55			m			
Mb	2.0+ 2001		Location	Location			
1	Operator 55						

Note that call functionality will be explained in chapter 'Basic call functionality' later in this document.

#### 3.1 Resizing the panels

You can resize each panel to the desired size by dragging the 'splitters' (marked orange below):

PeterCo	onnects				🛔 monica	(64111)	/ 😗 Av	ailable	~ 🗖	
Conve	ersations (F2)	Conversation detai	ls (F3)						(	3
Patrick Troughton (88039) RelipcoLogistics - Servicedesk		Patrick Troughton	Patrick Troughton (88039)							•
	© Destination: Sales department	8 Contactinformatie		Q	Contact not	:5			G	
	_	Detrick Transition								
Conto	(***)			(***	)					
Conto	cts (FO)	Contact details (F7	)						(	z
	Allison Price	Cathlin Schulz (8	8029) cs-Logistica					*		-
	Arman Jamesson	A Contact information		۹	Outlook cale	ndar				Q
-	Rethold Prestwood				Jan 21 – 25, 2	019 Month	Week Da	ay List y	$\langle \rangle$	
No.	BelpcoLogistics - Logistics	1 25			Mon 1/2	1 Tue 1/22	Wed 1/23	Thu 1/24	Fri 1/25	1
	Cathlin Schulz				6:00				1	
	Chaitan Brahney BelipcoLogistics - Logistics	Cathlin Schulz First name	Cathlin		7:00					
-	Charlotte Hudson	Last name	Schulz		0.00					

When the panels F2, F3, F6 or F7 have been made so small, that the buttons don't have sufficient space, the buttons will have a drop-down menu. To see the other buttons, click on the drop-down menu.

The positions of the splitters will be remembered between sessions.

The settings are stored on the Anywhere365 server, so you will have the settings available on other devices as well.

#### 3.2 Full screen mode

The menu bar contains a button for making the Receptionist full screen:



When the full screen mode is on, you can restore the screen to its original size with the restore button:



### 4 The contact list

In the contact list you can search for contacts, filter contacts and perform actions on the contacts that you have found.

#### 4.1 View modes in the contact list

#### 4.1.1 Default mode: Recently used contacts

By default the contact list will show the most recently used contacts. When a contact is used ( i.e. a call has been made to that contact and that call has ended, or a transfer to that contact has been done, or a mail has been sent to that contact, or a contact note has been saved for that contact) the contact is moved to the top position of the contact list.

Recen	tly used contacts (F6)	☆ 음+ ⊽ :
Q	ch by number, name, email address or department	ų.
•	PeterConnects Office	
sq	Sebastien Office France PeterConnects A365 Group - Sales	
*	Peter Pan Quality Assurance	: 🖻 📞
RP	Receptionist Paris 圓	
6	Jacqueline Portier	
<b>e</b>	Stephen Postma	
-	1	

#### 4.1.2 Favorites mode

When you click on the star icon in the title bar of the contact list, you will switch to the favorites mode. In this mode, you only see the contacts that you marked as favorite.

For information about adding, removing and sorting favorites, see paragraph "Favorites".

Favorite contacts (F6)	🗙 음+ 🖓 🚦
Q Bearch by number, name, empil address or department	ų.
Marc Derksen	: 🔼 📞
Jarno Holstein	
Bas van Leeuwen	
Norbert de Bruijn	

#### 4.1.3 Search mode

Whenever you type in a search term in the search box, or when you filter your contacts via the filter menu, the contact list will switch automatically to the search mode. In that mode the contact list will only show contacts that match your search query or your filter.

**Tip:** when search has completed, pressing ENTER will dial the top-most search result.

Search results (F6)	&+ V :
Q peter	×
Bas van Leeuwen	
Reema Apsingekar	: 🖂 📞
PeterConnects Office	
Sebastien Office France	
Peter Pan	
Dorette Verhoeven	

When you turn off the filter and clear the search box, the contact list will return to the previous mode.

#### 4.1.4 Title bar text depending on the mode

The title bar will show in text which mode you currently are working in:

When no filter is on and favorites are not set, the recently used contacts are shown, with the most recently used ones on top of the list:



When the 'Show favorites' button has been clicked, the favorite contacts are shown:



When a filter has been set, the filtered contacts are shown:



This title is shown, when you are searching for a contact, while a filter has been set:



This title is shown, when you are searching for a contact, while no filter has been set:



#### 4.2 Viewing the details of a contact

Click on any contact to see the contact details in the right screen.



#### 4.3 Multiple columns

The contact list supports multiple columns. When you drag the border of the contact list and increase the width, more columns will be displayed.

Recer	ntly used contacts (F6)								☆왕♡ :
Q	rch by number, name, email address or department								ų.
MS	Monica Schulz	٥	A wrong match 1	: 🔤 📞	2	ZcontactName0001	s	Stephen Mobieltje	
Ģ	John Prestwood	6	Operator 51		J5	John 52	ME	Monica 52	
05	Operator 52	J5	John 53		ME	Monica 53	05	Operator 53	
G5	George 54	J5	John 54		MS	Monica 54	05	Operator 54	
	Peter 55		Monica 57		No of the second	Operator 57	45	John 58	
MB	Monica 58	05	Operator 58		JĘ	John 59	ME	Monica 59	
JG	John 61	ME	Monica 61		06	Operator 61	JG	John 62	

#### 4.4 Contact types

2

Contacts are categorized, and this is indicated by the following icons:

Personal contact. This contact is only visible to you.



#### 4.5 Queues

As described above, queues are also listed as a contact in the contact list, so you can transfer calls to a queue:



#### 4.6 Presence

When the application is configured to show a contact's presence state, then the contact list might look like this:



Depending on the type of telephony system your company uses, the presence state may appear slightly differently. The colors however will remain the same. They are:

Green:	Available
Red:	Busy or Do Not Disturb
Orange:	Away
Grey:	Offline or unknown state

Hovering over the presence state indicator with your pointer will show the presence state in text. The presence state is also shown in Conversation details panel (F3) and in the Contact details panel (F7), see the next paragraph.

#### 4.7 Hook states

Just like the presence, also the hook states of your contacts are shown. The following hook states can be distinguished:



Idle: The contact is free.



**Connected:** The contact is currently busy in a conversation.



Hold: The contact currently has a conversation on hold.



**Ringing:** The phone of the contact is currently ringing.



**Dialing:** The contact is currently dialing a number.



**Do not disturb:** The contact has turned on the function "Do not disturb". This state overrides all the hook states above.



**Call Forwarding always:** The contact is automatically forwarding all calls. This state overrides all the hook states above. The number to which the calls are forwarded is shown in the tooltip:



When the contact has any of these hook states, it overrides the presence state. When you hover over the hook state, then you can see both states:



In the Contact details panel (F7) and Conversation details panel (F3) you see both the presence state and the hook state simultaneously:

Contact details (F7)	c ×
Operator Sixtyfive (68772)	le 💽 🛃 🔽 📞
2 Contact information	Q
First name	Operator
Last name	Sixtyfive
Primary phone number	68772
Alternative phone number	+31850068772

When you hover over either the presence or the hook state, you will see a corresponding tool tip:



#### 4.8 Actions in the contact list

When you hover over or select a contact, you have the following actions available:





Call the mobile number of the contact (only available when the contact has a mobile phone number)

In F6, the buttons "Edit Contact Note" 🗾 and "Add to Favorites" 📩 are always in the dropdown menu.



#### 4.9 Managing your favorites

#### 4.9.1 Adding a favorite

You can mark a contact as favorite by clicking on the star in the contact detail area:

Contact details (F7)		2 × 2	
Alan Jones (88003) III O Away O Idle			
2 Contact information	€	🗋 Calendar 🛛 😋	l
First name Alan		Week         Day         List         Oct 7 - 11, 2019         <         >         Today	

#### You can also use the favorites button in the contacts list (F6)



When the contact has been marked as favorite, a star will change color and become visible:

Contact details (F7)		c² ×	
Alan Jones <i>(88003)</i> Ⅲ ★ ● Away ● Idle ■			ן
2 Contact information	Q	🗋 Calendar	۹,
First name Alan	_	Week Day List Oct 7 - 11, 2019 < > Today	

#### 4.9.2 Removing a favorite

You can click on the 'Remove from favorites' button again, to remove the contact from your favorites.

#### 4.9.3 Displaying favorites

The contact list indicates your favorites with a star:



To view all your favorites, you can switch to the 'favorite mode' by clicking on the star in the title bar:



#### 4.9.4 Sorting favorites

When you are in the favorites mode, you can arrange your favorites by dragging and dropping a contact from one place to another. The dashed line indicates the place where your favorite will be moved.

Favor	ite contacts (F6)		🗙 🏹 🗄
Q Sear	ch for contects		ų
2	Marc Derksen 🛄 🚖		
	Sebastien Frabot Market A365 Group - Sales		
	Tim Field III 🌟 Sales	ice	
	Jarno Holstein III 🚖 Jdm Anywhere Connects - Development		
ŧ	Peter Pan III 🔶 Quality Assurance		
•	PeterConnects Office		: 🖻 📞
	Jacqueline Portier		

When there are multiple columns in the contact list, the dashed line will appear to the left or right side of the contacts, depending on whether you hover over the left or right part of a contact:



#### 1.1.1. Direct opening Manager contact info in F7 (Microsoft login only)

When clicking to manager name in a specific contact displayed in F7, the user will be able to see information or to interact with the manager of that specific contact. To return back, after opening the manager details, the user have to click to the contact name in F6

Stephen the King (+311079877	718)
Contact information	•
Contact Type	Corporate
First name	Stephen
Last name	King
Email	stephen.postma@pcadev.net
Primary number	+31107987718
Mobile number	+31623069416
Function	Senior Tester
Department	Quality Assurance
Organization	Anywhere365
Building	Anywhere365
Street of building	Helllingkade 9
Postal code of building	1234 AB
City of building	Maassluis
Country	Netherlands
State or province	Zuid Holland
Name of manager	PcaDev Admin 🔿 🔽 📭

#### **Direct opening Manager contact info**

Often being in contact with a customer and looking for a contact in F6 and this contact is not available one asks to know his manager or want to speak to this manager. With the possibility the of opening the manager contact information a call or transfer can be initiated from F7 that will show that contact information of the manager.



Contact details (F7)							
Bas van Leeuwen PcaDev						٢	٠
은 Contact Information	8	📛 Cale	endar				
Contact Type	Corporate	Week	Day List	7 Aug, 2	023 - 11 Aug.	2023	
First name	Bas	W 32	Mon 7/8	Tue 8/8	Wed 9/8	Thu 10/8	
Last name	van Leeuwen	01:00					
Email	bas vanleeuwen email@pcadev.net	02:00					
Function	Innovation Manager	03:00					
Department	Development						
Organization	PeterConnects	04:00					
Name of manager	PcaDev Admin	05:00					
	Show manager details	06:00					
		07:00					
		00.00					

#### 4.10 Searching and filtering

#### 4.10.1 Searching a contact

You can start typing to enter text in the search box to search in the contact list (you do not have to click into the Search field first). Your search text will be highlighted in the search results:



Select X in the search box or press "Escape" to clear the search text.

#### 4.10.1.1 Webex and Broadworks login

You can search on various fields, like Name, number, e-mail address or department. You can also search on parts a name. For example if a name is 'Elliot', it is possible to search on Ell', but also on 'iot'.

#### 4.10.1.2 Microsoft login

In case you use your Microsoft account to log in, all fields that you see in the contact information are searchable in the search box in F6.

**Note**: In this situation it is not possible to search on text that is not at the beginning of a word. For example, if a name is 'Elliot', it is possible to search on Ell', but not on 'iot'. When a name is 'Van Halen', it is possible to search on 'Hal'.

The contact list search results will be sorted based on the field in which the search term is found. From most important to less important the order is:

- 1. firstname
- 2. displayname and lastname
- 3. email
- 4. department
- 5. function
- 6. other fields.

This order of this list indicates the weighting of the search. If the searchtext is found in the same field then it will get more priority if it is an exact match. So when searching for "Rob" a person with firstname "Rob" will be shown first and a person with firstname "Robert" will be shown later. If the field and whether it is an exact match or not are equal then the favorites are shown first followed by the most recently used.

Note: A contact's displayname is defined in the User management of Microsoft 365 admin center.

When the result limit is reached, not all matching contacts are displayed. In such a case, enter a more precise search term to refine your search to find a specific contact.

#### *4.10.1.3 Searching for a number*

You can also search for a number. When the number is not found, you can still dial that number:



Instead of typing the numbers, you can also use the dial pad. You can toggle the dial pad on and off by selecting the dial pad II.

Search results	s (F6)			2+ 7
Q 2222				∝
	1	2 ABC	3 DEF	
	4 дні	5 JKL	6 MNO	
	7 PORS	8 TUV	9 wxyz	
	*	0	#	
Dial 222	22 n number			• % C

#### 4.10.1.4 Searching contact notes (only Microsoft login)

The contact notes can be searched now. You can do a specific search on with the prefix "note:" or a note specific search without the prefix.

Filtere	d search results (F6)	음* 🍸 :	Contact details (F7)	
Qr	note:test	× 🛯 🐺	Alain Deluxe (+31	(10) 59 27 892) '
AD	Alain Deluxe III 🛃 Jdm Anywhere Connects - Development		Contact information Contact Type First name	Corporate
<b>(</b>	Max Verstappen III Puality Assurance		Last name Primary number	Deluxe +31 (10) 59 27 892
E	echo		Function Department	Developer Development
8	My first group contact		Organization	Jdm Anywhere Connects
	group01		Note Test	
<b>C</b>	Peter Pan III PAnywhere365 - Quality Assurance			
J	Johan JJ. 't Hart 🗮 厚	8 2 4		

So when searching with a search term like "note:test", it will match contacts that have a contact note with a word that starts with (or equals to) "test".

When searching with a search term like "test" it will match the same contacts plus the contacts that have any other field with a word that matches "test".

A match on contact note will have the lowest priority so it is displayed at the bottom.

*4.10.1.5 Restriction of search results depending on license* 

Depending on your license, you might see an exclamation mark in the search box:



When you hover your mouse over the exclamation mark, you can see that your license might be restricting the number of group contacts that you see.



When using the Webex login, it is only possible to search contacts on the 'displayname' field in Webex. This field usually is a combination of first name and lastname.

A second limitation for the Webex login is that search is only possible when entering a search string that has a full match from the beginning of the name.

Example search on 'qa tech' returns the contact with name 'QA Tech':

Search results (F6)	음+ 7 :
Q qa tech	× 6 🏭
😪 🛎 🛨	: 💌 🔽

A search on only 'tech' will not return this contact, but returns results that match on the term 'tech' (in this case the email address):

Search results (F6)	음• 7 :	Contact details (F7)	
Q tech	× 🛯 👯	George Price (55131)	
John 53	: 💌 📞	Contact information	-
George Price		First name Last name	George Price
		Email	stephen@derkser <mark>.tech</mark>

#### 4.10.2 Filtering contacts from the filter menu

The contact list can be filtered in various manners. This feature is only available in the Premium and Advanced edition.

When you click on 'Filter settings', you can check by which item(s) you want to filter the contact list.



The filter settings will be displayed in a drop-down menu. The filter contains two sections: contact types and custom directories.



You can filter on one of the sections. For example, if you only want to show personal contacts, you check the checkbox in front of 'Personal contacts. Or if you only want to see Personal contacts and Queues, you check these checkboxes.

Or if you only want to see contacts that are in custom directory 'Pca', you can check this radio button.

You can also combine these sections. So you can indicate that you only want all Group contacts in custom directory 'Pca'. Or only your favorite corporate contacts.

Here is an example when you filter on Personal Contacts:

Filtered contacts (F6)	음+ 🕇	:
Q Search by number, name, email address or department	ų	
Debrah Orwell	:	
Jeff Healey		
Jip Doodle		
Max Baker		
Peter Connects		

When activating one of the filter options, the filter is automatically switched on, and it will remain active until you turn it off by clicking on the 'Filter contacts' button again.

Filtered contacts (F6)	음• 🔽 :
0	•••

When you turn on the filter again by clicking on the 'Filter contacts' button, the last used filter options will be applied again to your contact list.

	Recently used contacts (F6) ☆ 옵+ 🟹 :	
	• ···	
W	/hen the 'Filter Contacts' icon is like this: 🔽, the filter is active.	
W	/hen the 'Filter Contacts' icon is like this: 🔽, the filter is not active.	

Your filter settings will be remembered between your sessions.

#### 1.1.1. Contact notes are searchable

The contact notes can be searched now when logging in with a Broadworks account. You can do a specific search on with the prefix "note:" or a note specific search without the prefix. So when searching with a search term like "note:important" it will match contacts that have a contact note with a word that starts with (or equals to) "important". When searching with a search term like "important" it will match the same contacts plus the contacts that have any other field with a word that matches "important".

A match on contact note will have the lowest priority so it is displayed at the bottom.

Search results (F6)	음* 7 :	Homepage (F9)	
Q findcontactnote	× 🗈 🛱	<b>企</b> Homepage	
Peter Pan	: 🗷 🔽	Contact details (F7)	
		Peter Pan (71161)	Jole e
	I	S Contact information	=
	I	Contact Type	Corporate
		First name	Peter
	I	Last name	Pan
	I	Email	peterpan@pcadev.net
	I	Primary number	71161
	13	Mobile number	+31612345678
		Function	Senior Tester
	I	Department	Quality Assurance
	I	Organization	Anywhere365
	I	Building	Headquarters
	I	Street of building	Hellingkade
	I	Postal code of building	3144 EJ
		City of building	Maassluis
	I	Country	Nederland
	I	State or province	Zuid:Holland
		Note findcontactnote	

#### 4.10.3 Saving search queries

In the Receptionist you can save your search queries so you can use them later on. The saved search queries actually are a combination of a filter and/or a search string.

#### 4.10.3.1 Saving a search query

You can create a filter, whenever you are filtering and/or searching, for example when you are filtering on "Personal contacts" and you want all contacts that have "mon" in the name:

Filtered search results (F6)				&+ ▼	:
Q mon	<ul> <li>✓</li> </ul>	•	Personal contacts		

When you want to save this query, you can click on the "save" button in the search bar:

Filtered search results (F6)	음• 🕇 🚦
Q mon	×®₩
Monica Anderson	
Monica Brown	: 💌 🔽
Monica Dalton	
Monica Jason	
Monica Sewell	

**Note:** The save button can be disabled, if you have reached the maximum number of saved queries. In that case, you have to delete a saved search query first.

When you click on the save button, a popup will be shown, where you can give your saved query a name:
		×
🕼 Search query		
Name		
Search		
mon source:Personal		
	Clear	Save

In the popup, you can also see the Advanced Search text, that is used to list these contacts. (For more information about the Advanced Search syntax, see the paragraph about Advanced Search later on in this chapter).

Then click on button "Save", to finalize your search query. If you open your filter drop down, you will see that your search query has been saved. Right below the contact types, you will see a section that contains your saved search queries:



When you click on a saved search query, the contact list will show only the contacts that apply to that query. When you click on "(All contacts)", the filter will be cleared and you will see all contacts again.

# 4.10.3.2 Editing a saved search query

You can change the name of a saved search query by clicking on the edit icon  $\square$ . Then the popup will appear, where you can enter a new name.

# 4.10.3.3 Deleting a saved search query

You can delete a saved search query by clicking on the edit icon  $\square$ . Then the popup will appear. Click on the trashcan button to remove the name. A question will be shown if you want to delete the item, with two buttons "Yes" and "No". Click "Yes" to delete the item or click "No" to get back to editing the item.

# 4.10.4 Advanced search

The Receptionist that uses the standard login or the Teams login supports advanced search. This means that you can build your own search string that also includes filter options.

#### 4.10.4.1 Syntax

A search string contains keys and values. A key-value pair looks like this:

key:values

#### for example:

source:Personal,Corporate

In the example above, the search will be limited by only Personal and Corporate contacts.

A key-value pair of values cannot contain spaces, commas or a colon. If you need it to contain those, then the values need to be included in a set of quotes. For example:

key:"value 1","value:2"

#### 4.10.4.2 Overview of keys and values

<no key=""></no>	All values in the string that are not preceded by a key will be used as a search term. For example:
	dav
	will look for contacts that contain the word "dav" in either the first name, last name and some other fields. Another example:
	dav bow
	will look for all contacts that contain either the word "dav" and the word "bow" in the first name, last name, and some other fields.
Source	This will limit the search to the specified sources. For example
	source:personal
	will look for all personal contacts.
	The various sources are: - Personal - Group - Corporate - Queue

A combination of keys and values can be used for searching. For example:

dav source:personal

looks for all personal contacts and 'dav' in either the first name, last name and some other fields

It is possible to search the other contact details fields as well, the following fields are searchable:

Contact details field name	Example	Broadworks field name
First name	FirstName:dav	First name
Last name	LastName:dav	Last name
Email	Email:dav@dav.com	Email
Primary number	Phone:123	Extension
Alternative number	CompanyAltPhone:123	Phone Number
Mobile number	MobilePhone:123	Mobile phone
Pager	Pager:123	Pager
Title	Title:msc	Title
Organization *)	Organization:dev	Company name
Building *)	Building:dav	Office
Department	Department:sales	Department
Location	Location:dav	Location
Address of building **)	buildingAddress:dav	Address
Street of building *)	BuildingStreet:dav	Street address
Postal code of building	BuildingPostalCode:123	ZIP or postal code
City of building	BuildingCity:dav	City
State or province *)	BuildingState:dav	State or province
Country	BuildingCountry:dav	Country

\*) only available when using the Teams login.

\*\*) only available when using the standard login

For some fields in the contact details, a clickable filter icon appears next to the value, which inserts it as a search term, making it easy to find more contacts of the same kind:

Contact details (F7)	
Monica Schulz (	(55111) Idle
S Contact information	
First name	Monica
Last name	Schulz
Primary number	55111
Email	markovirtual@pcadev.net
Title	Mevr
Mobile number	0612345678
Location	first floor
Address of building	Nice street 1
City of building	Amsterdam
Postal code of building	1234AB
Country	Nederland
Department	Testing T
Note	
contact note added for monica	a

Following field shows this clickable filter icon:

- Function \*)
- Department
- Organization \*)
- Building \*)

\*) only available when using the Teams login.

### 4.10.5 Copy contact information

The Attendant Console provides functionality for easy copying contact information. Most of the contact fields present a <sup>10</sup> button when hovering over it. By clicking on this button the information in the field will be copied.

Contact details (F7)		
Peter van Baal   ANYWHERE365 (+31 (0) 10	250 3018)	
Contact information		
Contact Type	Corporate	
First name	Peter	
Last name	van Baal	
Email	peter.vanbaal@anywhere365.net	
Primary number	+31 (0) 10 250 3018	Сору
Function	Product Owner	
Department	Product Management	
Organization	Anywhere365	
Building	NL	
Street of building	Hal B & C Van Nellefabriek, Van Ne	elleweg 1

# 4.10.5.1 Copy information from the Notes

Often the additional Notes contain information, like telephone numbers that need to be copied. Copying this information is done by selecting the information that need to be copied and automatically it will copy the information.

Contact details (F7)		
Peter van Baal   ANYWHERE30	65 <i>(+31 (0) 10 250 3018)</i>	
S Contact information		
Function	Product Owner	
Department	Product Management	
Organization	Anywhere365	
Building	NL	
Street of building	Hal B & C Van Nellefabriek, Van Nelleweg 1	
Postal code of building	3044 BC	
City of building	Rotterdam	
Country	Netherlands	
State or province	Zuid-Holland	
Name of manager	Ruud Lendfers   ANYWHERE365	
Note		
The mobile number is: +31612345678		

#### 4.11 Editing contacts

#### 4.11.1 Adding a contact

You can add personal contacts to the contact list. The steps are:

1. Click on the 'Add contact' button in the title bar of F6:



2. Now, a pop-up appears in which you can enter the data of the new contact. Click on the 'Save' button to save the new contact. Click on 'X' to cancel.

Note that the fields in the image might differ depending on your edition of the Receptionist.

Personal contact	×
*Name	
*Primary phone number	
	Save

Another way to add a contact is:

1. Search for a number in the search box of the contact list.



2. Click on the 'unknown number' contact in F6. Then in the Contact Detail panel, you will see a button "Add" in the contact information widget. Click on that button.

You can also use the button in F6 to add the contact.

Filtered search results (F6)	Contact details (F7)
Q 0123456789 × 🖼 🗰	(0123456789)
Dial 0123456789         :         ☑           Phone number         :         ☑	2 Contact information
No results	Primary number 0123456789

3. Then the contact pop-up appears, see at the beginning of this paragraph.

# 4.11.2 Editing a contact

When you want to edit a personal contact, first search for the contact via the search box in the contact list and select the contact. Then click on the button "Edit" in the Contact Information widget in F7 or click the "Edit" button in F6 itself (Hidden behind the 3 dots).

= Search results (F6) 음· ▽ :	Contact details (F7)
Q <sub>james</sub> × 🖱 👯	James Summer (0123456789)
James Summer : S	Contact information
	Primary number 0123456789

The contact pop-up appears. When you are ready editing the contact, click on the button 'Save' to save the changes or on the 'X' to cancel editing.

Note that the fields in the image might differ depending on your edition of the Receptionist.

	×
Le Contact	
*Name	
James Summer	
*Primary number	
0123456789	
	Save

#### 4.11.3 Deleting a contact

To delete a personal contact, first search for the contact in the search box of the contact list and select the contact. Then click on the button "Edit" in the contact information widget in F7 or click the "Edit" button in F6 itself (Hidden behind the 3 dots).

Search results (E6) $0 + \nabla$ :	Contact details (E7)
Q <sub>james</sub> X 🗈 🠺	James Summer (0123456789)
James Summer : E C	Contact information      Image: Summer      Primary number      D123456789

A popup will open where you can edit the contact, but there will also be a trashcan button:

		×
4	Contact	
	*Name	
	James Summer	וך
	*Primary number	_
	0123456789	
$\langle$	Save	e

A confirmation message will be displayed, asking if you really want to delete this contact.

×
Yes No

Click on "Yes" to confirm to delete the contact or click on "no" if you don't want to delete the contact.

# 4.12 Shared contacts

Shared contacts are managed within the Receptionist, but contrary to Personal contacts, they are visible for all Receptionist users. The privilege to manage Shared contacts can be controlled from the Admin portal (see section: User privileges via Admin portal).

Note: Shared contact functionality is only available when logging in into the Receptionist with a Microsoft account.

# 4.12.1 Filtering shared contacts

You can filter on Shared contacts in the filter menu.



### 4.12.2 Adding a shared contact

You can add Shared contacts to the contact list in case your account has the rights to do so assigned in the Admin Portal. The steps are:

1. Click on the 'Add contact' button in the title bar of F6:



2. Now, a pop-up appears. Select 'Shared contact' and enter the data of the new Shared contact. Click on the 'Save' button to save the new contact. Click on 'X' to cancel.

Note that the fields in th	e image might differ depending on	your edition of the Receptionist
----------------------------	-----------------------------------	----------------------------------

	×
Le Contact	
O Personal	
*Name	
	83
Email	
*Primary number	
Mobile number	
Function	
Department	
Organization	
	Save

Another way to add a contact is:

1. Search for a number in the search box of the contact list. |



2. Click on the 'unknown number' contact in F6. Then in the Contact Detail panel, you will see a button "Add" in the contact information widget. Click on that button.

You can also use the button in F6 to add the contact.

Filtered search results (F6)	Contact details (F7)
Q 0123456789 × 🗈	(0123456789)
Dial 0123456789 E Sector Phone number	Contact information
No results	Primary number 0123456789

3. Then the contact pop-up appears, see at the beginning of this paragraph.

# 4.12.3 Editing a Shared contact

When you want to edit a Shared contact, first search for the contact via the search box in the contact list and select the contact. Then click on the button "Edit" in the Contact Information widget in F7 or click the "Edit" button in F6 itself (Hidden behind the 3 dots):



The contact pop-up appears. You can edit the contact, including changing it from Shared contact into Personal contact. You can also convert a Personal contact into a Shared contact and thus making it available for all users. When you are ready editing the contact, click on the button 'Save' to save the changes or on the 'X' to cancel editing.

Note that the fields in the image might differ depending on your edition of the Receptionist.

	×
L Contact	
○ Personal	
*Name	
Shared contact	
Email	
sharedcontact@test.com	
*Primary number	
+31987654321	
Mobile number	
Function	
QA engineer	
Department	
QA	
Organization	
Anywhere365	
Save	

#### 4.12.4 Deleting a contact

To delete a Shared contact, first search for the contact in the search box of the contact list and select the contact. Then click on the button 'Edit' in the contact information widget in F7.

Search results (F6)	≗• ⊽ ∶	Contact details (F7)	
Qjames	× 🗈 👬	James Summer (0123456789)	
James Summer	: 🔽 🕻	2 Contact information	James Summer
		Primary number	0123456789

A popup will open where you can edit the contact, but there will also be a trashcan button:

	×
L Contact	
O Personal   Shared	
*Name	
Can be deleted	
Email	
nolongerneeded@test.com	
*Primary number	
+1234567890	
Mobile number	
Function	
Department	
Organization	
	Save

A confirmation message will be displayed, asking if you really want to delete this contact.

	×
Le Contact	
*Name	
James Summer	
*Primary number	
0123456789	
Do you want to delete this item?	Yes No

Click on "Yes" to confirm to delete the contact or click on "no" if you don't want to delete the contact.

# 4.13 Contact Sources

In addition to the standard Contact Sources, the Attendant supports displaying and filtering contacts from additional sources such as Excel files on a Sharepoint/OneDrive environment. These Contact Sources are managed within the Admin portal.

The contacts are visible to all users that have access to the contact source itself.

### 4.13.1 Filtering contact sources

The configured contact sources are shown together with other contact types, in the screenshot below the Queues. The administrator can set up a custom icon for them. On the above screenshot there are 2 external contact sources configured: Excel Contacts and Star Wars Contacts. Using the checkboxes, any combination of contact sources can be enabled or disabled.



# 4.14 Sending an email to a contact

To send an email to a contact, press the email button in the contact list next to the person that you want to send the email to. This feature is only available in the Premium edition.

Recer	ntly used contacts (F6) ☆ 음+	V :
Qßear	rch by number, name, email address or department	÷
J	Johan 't Hart 🎚	
RA	Reema Apsingekar	L
РВ	PCa Bot	
TO	Test Complete	

Then the send mail popup will be shown. Here you can enter your email.

		P
Mail Template	Callback reminder (English)	•
Callback number	52111	•
Send to	any.one@anywhere365.com	
Subject	Please call back: Monica 52 52111	
Message	Missed call from: Monica 52 52111 Subject:	
	Kind regards, Operator 62	
	Email: operator62@bw.dev.onpeterconnects.com Phone: 62101	1

The form contains the following fields:

- Mail Template There are one or more mail templates configured. Each can be in another language. The first one is used as default. The mail template can be changed in the pop-up of the email.
- Callback number You can select a phone number the recipient should call back. The list contains the numbers from active calls as well as recent previous calls. The list of call back number is taken from the call history and can contain up to 10 numbers. After selecting a number the phone number and name of the caller (if known) are filled in, into the call back email template. See the screenshot below. The last selected call will be shown in the template.

Callback number	55131	•
Send from	55111	
Oraclas	55131	
Send to	(None)	
Subject	Please call back: George Price, 55131	

Send from The email address that will be used to send this email. This field cannot be edited in the Receptionist. For changing this email address, see chapter [Global configuration]. Note that this field is not shown when this has not been configured in the global configuration.

Send to	The email address of the contact. This field is prefilled when the email address of the contact is known. Otherwise, you can enter an email address.
Receive a copy	Check this if you want to receive a copy of the email in your own inbox. The email address that will receive a copy is shown next to the checkbox. If this field is not visible, please check if you have specified an email in your profile.
Subject	The subject of the mail. The subject will contain a predefined text with the data of the most recently selected call in the calls list (F2). You can change the text if needed.
Message	The content of the mail. The message will contain a predefined text with the data of the most recently selected call in the calls list (F2). You can change the text if needed.

Press button 'Send' to send the mail.

To cancel the email, you can press the 'X' button in the top right corner or press the 'ESC' button.

# 4.14.1 Opening the email form in a new window

Suppose you are writing a mail, but suddenly you have to do something in the Receptionist. Then you would like to close the mail popup and work on your mail later on. For this purpose the mail form has a pop out button:

		×
🔽 Send mail		
Language	English	•

When you click on this button, the mail form will be opened in a new browser window. Now you can switch between the browser windows to go to the Receptionist and back to the mail form later on.

When you have sent the email in the pop out window, a message will be displayed that you can close that browser window now:



# 4.15 Making a contact note

In F7 and F3 you can add a note to the selected contact by clicking on the 'Add note' button. This feature is only available in the Premium edition.

Contact details (F7)		c ×		
Arman Jamesson <i>(88015)</i>				
은 Contact informatio	on	Q	Contact notes	Q
First name	Arman		No notes	
Last name	Jamesson			

# You can also use the 'Add note' button in the contacts list (F6)

Bethold Prestwood BelipcoLogistics - Logistics	
Cathlin Schulz III 🔶 BelipcoLogistics - Logistics	

In the popup, you can enter the note. The notes can have a maximum of 1000 characters. Click on the 'Save' button, if you want to save the note. Click on the X button, if you want to exit the popup without saving the note.

🗹 Note	×
Arman does not work on Fridays	
	Save

Now, the note is shown in the contact notes widget and in the contact information widget. Be aware that the contact note can be seen by anyone, who has access to that same contact.

Contact details (F7)			ľ ×
Arman Jamesson (	(88015)		
S Contact information	୍	Contact notes	Q
First name	Arman		6
Last name	Jamesson	Note	ď
Business E-mail address	ArmJam@BelipcoLogistics.com	Arman doesn't work on Fridays:	
SIP-address	sip:ArmJam@BelipcoLogistics.co m		
Business primary phone number	88015		
Mobile number	7857626851		
Private phone number	2923543670		
Organization	BelipcoLogistics		
Department	Finance		
Building	London - EC2Y		
Note	ď		

Also, in the contact data in F6 and F7 it is indicated that there is a note for this contact. When you hover over this indication, you see the first 80 characters of the contact note.



You can edit the contact note in different ways:

- Click on the 'Edit note' button
- Click on the 'Edit' button in the contact information widget. The button appears while hovering over the widget
- Click on the 'Edit' button 🖵 in the contact note widget. The button appears while hovering over the widget

In the popup, edit the note and click on the 'Save' button. If you don't want to save the changes, click on the X button.

If you want to delete the note, click on the trashcan button in the popup:

🕑 Note	×
Arman does not work on Fridays	
	Save

You will be asked if you want to delete the item. Then click on 'Yes'.

🖍 Note	×
Arman does not work on Fridays	
	//
Do you want to delete this item?	Yes No

# 5 Call functionality

# 5.1 Connection with the server

When the Receptionist is starting up a connection with the server, you will see the following line of text in the conversations screen:

Conversations (F2)	l &
Connecting to the server	

When it takes too long to connect, the text will become accented and a "retry" button will appear. The button mentions how long it will take before the next retry attempt will occur. If you want to retry earlier, you can click on the "retry" button.

<u></u>

# 5.2 Call states, destination and call actions

In the screen below, there are three calls:

- A call on hold with Jasmine Abdel
- An active call (Elliot Amstell)
- An incoming call (Monica Jansen)

Conv	ersations (F2)	৾
	Jasmine Abdel (88000) BelipcoLogistics - Servicedesk © 0:13 - On hold © Destination: main queue	থা 🕨 🥌
	Elliot Amstell (88001) BelipcoLogistics - Servicedesk © 0:12 - Connected © Destination: main queue	୩ 🔳 🧖
Ø.	Monica Jansen (+31105973111) © 0:20 - Incoming call © Destination: main queue	

The following paragraphs describe the elements you can see in the calls in the Conversations (F2) panel.

### 5.2.1 Call states

The state of the calls is indicated in a small circle in the avatar of the other party.

- This is the active (connected) call
- This call is on hold
- The caller is hearing a pre-alert announcement

#### 5.2.2 Destination information

A call can show the following information about the original destination of the call:

O Destination:	This is the destination of the call
→ Fallback from:	This shows from which destination the call falls back into Conversations
→ Forward (always) from:	This is shown when somebody is calling to a contact, who has set his call forwarding to you or to the queue, that you have

joined/monitored. Note that the forwarding information is only shown, when the call is in the My Calls section of the Conversation panel F2. In the parentheses you can see the reason of this forwarding: Always: The call is always forwarded • Busy: The call is forwarded when the agent is busy • No answer: The call is forwarded in case of no answer • Unavailable: The call is forwarded because the agent is • unavailable Handled by: This is shown in a call when the operator has picked up this call before.

### 5.2.3 Call actions

Depending on the state of the call, you can perform the following actions:



# 5.2.4 Greyed out pickup button

When the attendant is only using a calling application and no physical device, the pickup button for an incoming call will show up as greyed out, indicating that the phone client should be used to pick up the call.

Conversations (F2)	👁 🌲   😵	Homepage (F9)
℅ My calls (1)		<b>G</b> Homepage
Peter 55 (55141) 1:07 - Incoming call Destination: Operator Sixtyfive	Please pickup this call u	sing your calling application
③ Support (0) ∨		

# 5.3 Directed call pickup

Via the contact list, you can pick up a call that has been made to one of the other colleagues. For example, when someone is calling Allison Price, and she's not at her desk, you can pick up the call here:



**Technical note:** On the telephony system, the receptionist must have the *Directed Call Pickup* service assigned.

# 1.1.1. Calling Enterprise contacts from different groups with same extension

When you are part of an enterprise Broadworks deployment, you are able to find users from all groups and call them

# 5.4 Transferring a call

There are several ways to transfer a call:

# 1. Transfer an active call to a hold call (announced transfer)

When you have an active call and a call on hold, you can click on the transfer button to transfer the active call to the held call:

Conv	versations (F2)	্০
	George de 65 <i>(65131)</i> • 1:20 - On hold • Destination: Operator 65	গ 🕨 🗖
<b>V</b>	John van der 65 <i>(65121)</i> <b>5</b> 1:02 - Connected O Destination: Operator 65	

# 2. Transfer an active call directly to a contact (blind transfer)

When you have an active call, you can search in the contact list and transfer the call to that contact:

Conv	ersations (F2)	્૦ે
	Jasmine Abdel (88000) BelipcoLogistics - Servicedesk © 2:59 - Connected © Destination: Sales department	
Conte	acts (F6)	
Q 880	002	×
	Tom Atkinson BelipcoLogistics - Reception	গা 🖂 🕓

# 3. "Drag and drop" a call to a contact (blind transfer)

When you have an active call, you can drag that call to a person in the contact list. When dragging, you will see dotted lines where you can drop the call.

Conv	ersations (F2)	<mark>0</mark> ې
	Jasmine Abdel (88000) BelipcoLogistics - Servicedesk © 0:15 - Connected O Destination: Sales department	
Conto	acts (F6)	
Q 880	002ine Abdel <i>(88000)</i> portogistics -Servicedesk osTom Atkinson Belipcotogistics - Reception	ଆ ଲ × ୧୩ ଲ 🕻

# 5.5 Call recording

When you are in a call, there is the possibility to record the call or fragments of the call. To start recording, click on the red record button:

Conv	ersations (F2)	را
	Jasmine Abdel (88000) Belipcol.ogistics - Servicedesk 1:47 - Connected O Destination: Sales department	

The recording button will start flashing quickly, while the recording is starting up. When the recording has started, the pause and stop button will appear too, and the recording button (disabled) will start flashing slowly, so you know that a recording is in progress.

Press the pause button to pause the recording. After pausing you can resume by clicking on the start button again.



Click the stop button to stop the recording. After stopping the recording, you can start a new recording by clicking the start button again.

Conv	ersations (F2)	و
	Jasmine Abdel (88000) BelipcoLogistics - Servicedesk • 1:47 - Connected ② Destination: Sales department	

# 5.6 Call recording options

The administrator can set some call recording options. Depending on the chosen option, the relevant recording buttons will be visible and behave accordingly. The following options are available:

- **Always:** When a conversation starts, the recording will automatically begin. There will be no stop or pause button. The recording button will indicate that the recording is in progress.
- **Always with pause/resume:** When a conversation starts, call recording will automatically begin. But you can press the pause button to stop the recording temporarily.
- **On demand:** The conversation is always recorded, but the recording will be deleted after the conversation has been finished. However, before or during a conversation, you can indicate (by clicking on the recording button) that the recording must be saved. Once this indication has been given, the recording can be paused and resumed.
- **On demand with user-initiated start:** During the conversation, the receptionist can start, pause and stop the recording.

Never: Conversations cannot be recorded; the recording buttons will not be visible. -

The receptionist does not show, which option has been selected. This can be indicated by the administrator.

Please note that pressing F5 or re-login is necessary, when the recording options have been changed by the administrator, while the user has still been logged in.

# 5.7 Conference calls

**&**+ You can set up a conference call by clicking on the 'Add to current conversation' button when there are at least 2 connected calls.

You can add a new call to a conference call by clicking on the 'Add to current conversation' button.

Conversations (F2)	্
Elliot Amstell (88001) BelipcoLogistics - Servicedesk © 0:12 - Connected © Destination: main queue	
Conference Call (+3 attendees) Currence Call (+3 attendees) Curr	

Conference calls can be identified by the 'conference' icon:

A conference call has the following action buttons:



Put the conference on hold. The other members of the conference can still talk, but you will be temporarily out of the conference.



Retrieve the conference call from hold (only when it is currently on hold).



Hang up. This will end the entire conference call.

The conference call contains a small tab for every member of the conference call. When you click on such a tab, the name of that member is displayed in the Conversation details panel (if the name of that member is available).

You can apply the following actions on any member:



Mute the conference member. This way the member can still hear the conference but cannot be heard by other members anymore. When a conference member is muted, the icon is shown red:



Click on the icon again to unmute the member.



Deaf the conference member. This way the member cannot hear the conversation anymore. When a member is deafened, he is also automatically muted.

When a conference member is deafened, the icon is shown red:



Click on the icon again to undeafen the member.



Remove the member from the conference. This action cannot be undone.

# 5.8 Working with queue calls

When you have chosen to join or to monitor queues, the display of the calls will be different. Queue functionality is only available in the Premium edition.

NB: For more information about joining and monitoring queues, see the paragraph about Queue Management in chapter [Personalizing the Receptionist].

Conve	ersations (F2)	🧶 🖉 🍠	
J My cal	s	1	
M6	Monica 64 (64111) © 021 - Incoming call © Destination: Operator 64		My calls
6	Postma 51 (51101) % 016 - Outgoing call		
A Prima	ry Queue 53 (1) 🔺		1
M6	Monica 64 (64111) C 3:29 - Incoming call S Destination: Call Center Primary Queue 53		Calls on joined queues
Secon	idary Queue 53 (0) 🔺		
Prima	ry Queue 54 (0) 🔥	1	
👁 Tertia	ry queue 53 (1) 🔸		Calls on
J6	John 64 <i>(64121)</i> © 029 - Incoming call © Destination: Call Center Tertiary queue 53		monitored queues

When you have selected queues to join or to monitor, the calls will be divided by headers:

- **A header named 'My calls'**. These are the calls that are on your phone. (The calls that you make yourself, the calls that are directly made to you and the calls that are dispatched to you from a queue)
- A header for each joined queue
- A header for each monitored queue

# 5.8.1 Toggling visibility of all joined or all monitored queues

You can toggle the visibility of all joined queues simultaneously by clicking the bell icon in the F2 header.

Conversations (F2)	و 💁 🗞
J My calls	

You can toggle the visibility of all monitored queues simultaneously by clicking the eye icon in the F2 header.

Conversations (F2)	🧶   🎐
J My calls	

When both the joined and monitored queues are toggled off, the Conversations panel will only contain your direct calls. In that case the division into my calls/joined queues/monitored queues is no longer relevant and will be turned off.

Conversations (F2)	🧶 🔦

# 5.8.2 Toggling visibility of a single queue

The calls in each queue can be made visible/invisible by clicking on the expand/collapse arrow in the queue header.



### 5.8.3 Restriction in picking up queue calls

There is a limitation in picking up queue calls. One user can pick up a maximum of 5 calls concurrently. Therefore, when you try to pick up the 6<sup>th</sup> queue call, you will get a message like shown below:

You can handle a maximum of 5 calls X on your device. Please disconnect one of your other calls first before answering this call.

# 5.9 Calling or transferring to other numbers of a contact

### 5.9.1 Calling to other numbers of a contact

The contact information widget shows all phone numbers of a contact. For example the primary phone number, the mobile number etc.

The phone numbers are underlined to indicate that they are links. When you hover over a row that contains the phone number, a small dial button will appear. When you click on this dial button, a phone call will be made to that phone number.

Contact details (F7)		
Cathlin Schulz (88029)	5	
8 Contact information		Q
First name	Cathlin	
Last name	Schulz	
Primary phone number	88029 🕓	
Email	schulz@belipco.com	
Note		ď
Absent on Wednesday		

When you click on the number itself, a pop-up will be shown, asking if you want to make a call to the specified phone number:



Press on 'Dial' if you want to make the call and on 'X' if you don't.

# 5.9.2 Transferring to other numbers of a contact

When you want to transfer a call to another number of a contact, you select that contact in the contact list and go to the contact information widget in the Contact Details. When you hover over a row that contains a phone number, a small transfer button will appear. When you click on this transfer button, the call will be transferred to that phone number.

Contact details (F7)	
Monica Testcomplete (85111)	
Contact information	
First name	Monica
Last name	Testcomplete
Primary phone number	85111
Email	monicatestcomplete@pcadev.net
Title	mevrouw
Mobile phone	0612345678
Adress	Hellingkade
City	Maassluis
Zip code	3144 EJ
Country	Netherlands

When you click on the number itself, a pop-up will be shown, asking if you want to make a call or a transfer to the specified phone number:



Press on 'Dial' if you want to make the call, on 'Transfer' if you want to transfer the call and on 'X' if you want to do nothing.

# 5.10 Selecting a device for calls

It is possible to select a device as primary device for answering calls and starting calls. In order to do that, you can click on the drop down in the main menu bar and select the device you want to use.



The selected device will be remembered between sessions. When you sign in in the Receptionist and your previously selected device is not available, the first device in the list will automatically be selected, and a message will be shown with the new auto-selected device.

When a new device is signed in or a device is signed out, this will not be detected automatically by the Receptionist. In order to see the added or removed devices, you need to sign in again or refresh the web page in which the Receptionist runs, in order to make the changes effective.

**Note:** At this moment, devices that are used on a mobile phone or tablet are not included in the list of devices.

# 6 Receptionist status

### 6.1 Presence

In the right top corner of the Receptionist, you can see your own states. Default you will only see your presence state:

All (All devices)	Available	Operator PeterConnects (0854863347)	• Unknown	

It is possible to change the state by clicking on the three dots next to the state, or on the state itself. This will open a dropdown menu where you can select other possible states. The state will be changed directly after selecting a different state.



# 6.2 Do not disturb

When you have "Do not disturb" switched on, it will be indicated by the do not disturb sign as follows:



For switching the "Do not disturb" setting on or off, see the paragraph about Conversation settings in chapter [Personalizing the Receptionist].

#### 6.3 Forward all calls

When you have "Forward all calls" switched on, it will be indicated with the forward sign as follows:



When you hover over the forward sign, it will display to which number all calls are forwarded.



For switching the "Forward all calls" setting on or off, see the paragraph about Conversation settings in chapter [Personalizing the Receptionist].

**Note:** When you have both "forward all calls" and "do not disturb" switched on, only the "forward all calls" state will be shown.
# 7 Personalizing the Receptionist

You can open the "Preferences" dialog from the menu bar:

L operator65(	۵bw.dev.onpeterconnects.com (0850068772)	🔮 Available		•
🗘 Prefer	ences			~
i Help			قا ا	3

In the preferences dialog, there are several tab pages on which you can set your own personal preferences. These tab pages will be described in the next paragraphs.

#### 7.1 Interface preferences

In the preferences dialog, on the Interface tab, you can select a layout, a theme, the date and time format, some notification settings and the language:

Preferences			
🖵 Interface 🦊	🕻 Group 🥄 📞 Conve	ersation	
Layout		Theme	
& &	L L L	<ul> <li>Default</li> <li>Dark</li> <li>High contrast</li> </ul>	
2 2 2 2	L C C	, Date and Time forme	at
Tabbed layout		Date format	DD MM YYYY 🗸 🗸
<u> </u>		Date delimiter	•
		Time format	24 Hour 👻
Notification		Language	
Notification limit	5	Display language	English 👻
	5		
Notification duration (s)			

## Layout

There are currently five layouts to be chosen, ranging from one to three columns. In these layout settings, the icons mean the following:

Conversation area (F2)
 Conversation detail area (F3)
 Contact list (F6)
 Contact detail area (F7)

There is also an option for the tabbed layout. In this case, the Homepage, Conversation details and Contact details are displayed in tabs.

Homepage (F9) Conversation detail	ls (F3)	Contact details (F7)	ď
Operator Sixtyfive <i>(68772)</i>			-
은 Contact information	<del>Q</del>	Contact notes	Q
First name Operator		No notes	
Last name Sixtyfive			

#### Theme

You can also select a theme: The default theme is shown from the start. The High Contrast theme can be used when you have difficulty seeing contrast. The Dark theme can be selected if you prefer a less bright screen.

#### Date and time format

The way that dates and times are shown can be customized by selecting the wanted date and time format and the date delimiter.

#### Notification

On the bottom of this screen you can select a location for popup notification messages and how many you will get maximum at one time. A notification can look like this:



#### Language

You can set the display language to English, Dutch, French or German.

## 7.2 Conversation settings

On the tab "Conversation" you can configure some settings, that have to do with telephone conversations. The settings are described in the following paragraphs.

🖵 Interface 🕻 Conversa	ition R Queues	Availability	
Call handling			
Call waiting 🛛			
Auto answer direct calls			
Busy camp on			
Enable busy camp on: 2		120	Seconds
Drahihit transfer to voice mail on bli			
Prohibit transfer to voice mail on bil	nd transfer		
Prohibit transfer to voice mail on bit	nd transfer insulted transfer		
Prohibit transfer to voice mail on bit	nd transfer		
Prohibit transfer to voice mail on bit	nd transfer		
Prohibit transfer to voice mail on bit Prohibit transfer to voice mail on co Audio alerting My Calls	nd transfer insulted transfer Classic /	Alert 1	•
Prohibit transfer to voice mail on bit Prohibit transfer to voice mail on co Audio alerting My Calls Joined queues	nd transfer Insulted transfer Classic / Classic /	Alert 1 Alert 2	•
Prohibit transfer to voice mail on bit Prohibit transfer to voice mail on co Audio alerting My Calls Joined queues Monitored queues	nd transfer Insulted transfer Classic / Classic / Classic l	Alert 1 Alert 2 Beep	•
Prohibit transfer to voice mail on bit Prohibit transfer to voice mail on co Audio alerting My Calls Joined queues Monitored queues Suppress alerts when in a call	nd transfer Insulted transfer Classic / Classic / Classic l	Alert 1 Alert 2 Beep	•

### 7.2.1 Call handling



Call Waiting allows you to put your current call on hold to answer a different incoming call.

When you turn on 'Auto answer direct calls', incoming direct calls (i.e. joined queue calls and calls to the operator directly, not queue calls that not have been joined) will automatically be picked up for you. A direct call is only picked up, if you don't have a current connected call yet. When you hang up your current connected call, the next incoming direct call will automatically be picked up.

Note that Auto Answer only works, when you are logged in into the Receptionist.

#### 7.2.2 Busy camp on

Busy camp on			
Enable busy camp on:	120	Seconds	
<ul> <li>Prohibit transfer to voice mail on blind transfer</li> <li>Prohibit transfer to voice mail on consulted transfer</li> </ul>			

This feature allows you to camp on a call to a busy destination. You can configure the time before the call will fallback. This feature is only available in the Premium and Advanced edition.

You can also indicate whether it must be prohibited to transfer to voice mail.

#### 7.2.3 Audio Alerting

In this section you can set audio alerting. When audio alerting has been set, an audio signal is played when a new call comes in. The type of alerting sound can be chosen and can be different for queue calls and my calls. It is also possible to suppress alerts when you are in a call, using the checkbox 'Suppress alerts when in a call'.

An alternative audio output can be set as the 2nd ringer device. Which is specially convenient when using ACS. The "Test second ringer" will test the audio which is set for the "My Calls".

Setting the "2nd ringer device" to "None" will play all audio (voice and ringer) on the selected audio output of your computer.

My Calls	Classic Alert 1	
Joined queues	Classic Alert 2	
Monitored queues	Classic Beep	
Suppress alerts when in a call		
2nd ringer device	Default - Headphones (Intel® Smar	t Sound Tech
Test second ringer		

**Note** :Be aware that additional consent needs to be provided to your browser to get access to the microphone when opening the preferences.

PeterC	peterconnects.bw.acc.onpeterco × nnects.com wants to			<b>* •</b> (
Conversatic & My calls (0) A Service	Use your microphones	. الم	Homepage (F9)	➢ Preferences ☐ Interface
			<ul> <li>&gt; (→ 085071203</li> <li>&gt; (→ Bud Wiser</li> </ul>	

After that it will show red notification the browser tab showing that you audio being set the attendant.

#### 7.2.4 Calling timeout



In the section "Calling timeout", you can set the amount of seconds for the on hold timeout. When the on hold time of a call exceeds the amount of set seconds, the line of the call that shows the on hold time and status starts blinking.

## 7.3 Queue management

🖵 Interface 🕻 Conversation	Level 2 Availability	
oined queues		
ueues preselected for call distribution	Secondary Queue 53, Queue1	
elect queues to join for call distribution	TestQueue3 × Service Desk ×	××
itial ACD state after sign-in	Available	•
CD State after sign-out	Sign-Out	
urrent ACD state	Available	
ou can select a maximum of 10 queues for call c	listribution	
Ansitored autouss		
elect queues to monitor		
	Sales Department queue × ba TestQueue3 ×	sicCallCenter × × ×
ou can select a maximum of 10 queues to monit	or	

Queue functionality is only available in the Premium edition.

#### 7.3.1 Joined queues

To add a queue to join, open the drop-down box (with the arrow pointing down) and check which queues you want to join. To remove a queue, click on the X behind the queue or uncheck the queue.

You can select a maximum of 10 queues to join.

When you have selected a queue to join, phone calls on that queue will also ring directly on your phone, when it is dispatched to you.

Some queues might be preselected for you by an administrator. These will be displayed as "Queues preselected for call distribution".

#### 7.3.1.1 ACD state

For the joined queues, you can set the initial ACD (Automatic Call Distribution ) state after sign-in, the ACD state after sign-out and the current ACD state:

Initial ACD state after sign-in	Available 🗸
ACD State after sign-out	Sign-Out 🗸
Current ACD state	Available 🗸

The initial ACD state after sign-in is the ACD state directly after login into the Receptionist. You can configure it to be set on:

- <Leave unchanged> (do not change after login)
- Available (change it to 'Available' after login; this is the default behaviour)
- Unavailable (change it to 'Unavailable' after login)

Initial ACD state after sign-in	Available 🗸
ACD State after sign-out	<leave unchanged=""> Available</leave>
Current ACD state	Unavailable

The ACD state after sign-out is the ACD state after you have logged out. You can configure this to be set on:

- <Leave unchanged> (do not change after log out)
- Available (change it to 'Available' after log out)
- Unavailable (change it to 'Unavailable' after log out)
- Sign-Out (change it to 'Sign-Out after log out; this is the default behaviour)

ACD State after sign-out	Sign-Out
Current ACD state You can select a maximum of 10 queues for call distribution	<leave unchanged=""> Available Unavailable Sign-Out</leave>

With the current ACD state you can change your ACD state temporarily during your session.

You can configure your current ACD state to:

- Available
- Unavailable



The current ACD state is also shown in the main menu of the Receptionist:

All (All devices) : Available Sperator 64 (64101)	🗞 Available 🗖
	ď
All (All devices) Lunavailable Operator 64 (64101)	🗞 Available 🗖
	Ľ

When your ACD state is 'Available', incoming calls on the joined queues will also ring on your phone.

When your put your ACD state to 'Unavailable', incoming calls on the joined queues will not ring on your phone.

#### 7.3.2 Queues to monitor

With a similar drop down as for selecting joined queues, you can also select which queues you want to monitor.

Primary Queue 53 × Tertiary queue 53 × Secondary Queue 53 ×
(Select All)
Primary Queue 53
✓ Secondary Queue 53
Primary Queue 55
✓ Tertiary queue 53

You can select a maximum of 10 queues to monitor.

**Note:** When you have chosen to monitor a queue, but you have not joined it, phone calls on that queue will only be visible in your Receptionist application.

## 7.4 Availability

On the tab 'Availability' you can set the availability of the receptionist, regarding 'Do not disturb' and 'Forward all calls'.

<b>幸</b> Preferences	×
🖵 Interface 💊 Conversation	Queues Availability
, Do not disturb	
Activate do not disturb	
Ring splash 🛛	
Forward all calls	
Forward all calls to number 🔮	0851122290
Ring splash 🔮	

## 7.4.1 Do not disturb

By activating 'Do not disturb', you can block all incoming calls. If configured, they will be forwarded to voice messaging. This feature is only available in the Premium and Advanced edition.

The 'Ring splash' setting is to play a notification on your phone as a reminder that "Do not disturb" has been activated.

## 7.4.2 Forward all calls

This setting is for automatically forwarding all calls to a specified number. This feature is only available in the Premium and Advanced edition.

The 'Ring splash' setting is to play a notification on your phone as a reminder that "Forward all calls" has been activated.

#### 8.3.3 Dynamic queue ordering

An option in the preferences is added to enable dynamic queue ordering in F2. There are 2 options:

- Default: no change in existing behavior, so no dynamic sorting
- Sort queues dynamically

<b>≋</b> Preferences	×
🖵 Interface 🕻 Conversation 😤 Qu	Jeues 💄 Availability
Joined queues	
Select queues to join for call distribution	(Select one or more) Service Desk, Service Desk queue 2
Initial ACD state after sign-in	Available
ACD State after sign-out	Sign-Out 👻
Current ACD state	Available
You can select a maximum of 10 queues for call distribution	n
Monitored queues	
Select queues to monitor	(Select one or more) BasicCallCenter, Testcomplete Monitored, Que
You can select a maximum of 10 queues to monitor	
Queue Order	
<ul> <li>Default</li> <li>Sort queues dynamically. The queue with the longest w</li> </ul>	raiting call will be on top

Business rules:

- Joined Queues are always shown above the Monitored queues
- Longest ringing call in a joined queue will show on top.
- Longest ringing call in a monitored queue will show on top of the monitored queues list.

For the receptionist user it looks like this. No calls:

Conversations (F2)	
𝗞 My calls (0)	
✓	
✓	
✓ ⊚ BasicCallCenter (0)	
✓ ⊚ Testcomplete Monitored (0)	
✓ ③ Queue2 (0)	
💛 💿 Primary Queue 58 (0)	

Call on Joined queue 'Sales department queue', queue moves to top.

Conversations (F2)	o 👌 😵
🗞 My calls (1)	
SP StephenOnNet Postma (5555) C 0:02 - Incoming call Destination: Sales Department queue	
A Sales Department queue (1)	_
StephenOnNet Postma (+31623069416) 0:02 - Incoming call Destination: Sales Department queue	<b>.</b>
ン ら Service Desk (0)	
ン	
<ul> <li>SasicCallCenter (0)</li> </ul>	
<ul> <li>S Testcomplete Monitored (0)</li> </ul>	
M (10)     M	

Call incoming on Monitored queue 'Primary Queue 58', queue moves to top of monitored queues

Conversations (F2)	I 🕹 😓
℅ My calls (0)	
ン	
A Service Desk queue 2 (0)	
<ul> <li>Primary Queue 58 (1)</li> </ul>	
SP StephenOnNet Postma (+31623069416) 0:04 - Incoming call Destination: Primary Queue 58	
✓ ⊚ BasicCallCenter (0)	
<ul> <li>Testcomplete Monitored (0)</li> </ul>	
V @ Queue2 (0)	

# 8 Keyboard navigation

The Receptionist allows keyboard navigation via shortcut keys. This will help keyboard centric operators to operate faster.

The shortcuts are shown in the Help menu:



· · ··································		
I Shortcuts	Application messages 🛛 What's new 🕚 License	
Global shortcuts		
F2	Activate conversations panel	
F3	Activate conversation details panel	
F6	Activate contacts panel	
F7	Activate contact details panel	
F9	Activate Homepage panel	
F4	Put current conversation on hold	
Alt + P	Add contact (Alt+P)	
Alt + R	Show most recently used (Alt+R)	
Alt + V	Show favorites (Alt+V)	
Alt + L	Turn filter on or off	
Alt + K	Show filter settings	
Tab	Navigate forward	
Shift + Tab	Navigate backward	
Shortcuts for conversations panel (F2)		
-	Show conversation details in E2 / Show mana calls	
2	Hite queue calls	
*	Select previous conversation	
	Select previous conversation	
+ Fotor	Dickup conversation / Petrieve cell from held	
Citral & Entran	Fickup conversation / Retrieve can iron hold	
Dalata	Hanava	
Ctrl + Delete	Force hang up call	
Shortoute for conversation details name	((53)	
shortcuts for conversation details parte		
Enter	Pickup conversation / Retrieve call from hold	
Ctrl + Delete	Force band up call	
Shortcuts for contacts panel (F6)	N	
$Ctrl + \rightarrow$	Show contact details in F7	
-	Show contact details in F7 (with one column in F6)	
<b>→</b>	Move to the right in the contact list (with multiple columns in F6)	
e-	Move to the left in the contact list	
Ť	Move up in the contact list	
1	Move down in the contact list	
Enter	Dial or pickup	
Escape	Cancel search	
Ctrl + Enter	Blind transfer	
Ctrl + E	Send mail to selected contact	
Shortcuts for contact details panel (F7)		
Enter	Dial or pickup	
Ctrl + E	Send mail to selected contact	

## 9 Widgets

The detail areas of the Receptionist (and the homepage as well) contains widgets. These are small application parts that operate on either the currently selected conversation or the currently selected contact.

### 9.1 Maximizing and restoring

You can maximize a widget within its containing panel by clicking on the maximize button:

Contact details (F7)							12° ×
Operator 51 <i>(51101)</i>							* • •
2 Contact information		😑 📋 Cale	ndar				
First name	Operator	Week	Day List	Feb	15 - 19, 202	1	< > Today
Last name	51	W 7	Mon 2/15	Tue 2/16	Wed 2/17	Thu 2/18	Fri 2/19
Primary phone number	51101	06:00					
Email	stephen.postma@pcadev.net	07:00					
		08:00					
		09:00					
		10:00					
		11:00					

And you can restore the widget again by clicking on the minimize button:

ontact de	etails (F7)				2 ×
Ope	erator 51 <i>(51101)</i>				
Calendar					-
Week Day	List		Feb 15 – 19, 2021		< > Today
W 7	Mon 2/15	Tue 2/16	Wed 2/17	Thu 2/18	Fri 2/19
06:00					
)7:00					
8:00					
9:00					
10:00					
1:00					

Note that the maximize button is not shown when there is only one widget in the panel, because when there is only one widget, it is already maximized to the contents of the panel.

### 9.2 Moving the widgets around

When you want to change the layout of the widgets, you can click on the 'Edit widgets' button. The widgets become editable and you can drag and drop them and resize them. The widgets are positioned on a grid that becomes visible when dragging / resizing:

Contact details (F7)	=	
Operator 51 (51101)		
2 Contact information	×	🗋 Calendar 🗢 🗴
오 Contact information		법 Calendar

The widgets will automatically snap to the closest line on the grid. You can give the widgets any size and position.

You can automatically organize the widgets again by clicking on the 'Arrange widgets'

You can remove a widget by clicking on the 'X' (Remove widget) in the title bar of the widget.

You can stop editing the widget by clicking on the 'Stop editing'	$\mathcal{Z}$	button.
---	---------------	---------

Note that the content of the widgets is not shown in edit mode. The content will be shown again when you stop editing the widgets.

\_\_\_\_

#### 9.3 Adding a widget

In the edit modus of the widgets, click on the button 'Add a widget':

Then the widget library will be shown, where you can select a widget:

**Note:** The actual widgets that are available to you might be different than the ones displayed below. This image is just for demo purposes.

	Widget	t library	
🛗 Calendar	<mark>ես։</mark> Chart	Contact information	Contact notes
つ Conversation History	<b>Q</b> Web page		
			Cancel

Select a widget and click on the button "Add". After adding a widget, you can configure it.

Settings for widget 'Contact notes'
Title (leave empty to use default title 'Contact notes')
Icon Select

At least you can edit the title and the icon. The title is not mandatory. If you leave the title empty, the default title will be used. This default title will also automatically be translated, when you select another language via 'Preferences'. When you do enter a title, that title will override the default title and will not automatically be translated, when another language is selected.

If you want to change the icon, you can click on the 'Select' button to show the icon selector. Then you can search for a specific icon, for example 'note'.

Settings	s for widget 'Contact notes'	$(\times)$
Title (	(leave empty to use default title 'Contact notes')	
Icon		
G	Select	1
	note	
	U	ок

The icon search is only available in English, because the icons are coming from a public library named 'Font Awesome'.

Some widgets require more configuration. Those will be described in the next paragraph.

Click on 'OK' to add the widget. Click on 'X' to exit the settings screen.

You can change the configuration of widgets that had already been added by clicking on

the 'Configure widget' button 🌄.

Exit the editing of the widgets by clicking on the 'Stop editing' button 🖉.

#### 9.4 URL Placeholders for widgets

Widgets that need a URL can have placeholders. These placeholders will be used by the Receptionist to provide the target data provider with extra request data. The placeholders are:

- [CLI] This placeholder stands for "Calling Line Identification". This is the phone number of the person that is calling.
- [EMAIL] The Receptionist uses this tag as placeholder for the email-address of the currently selected person. For example, if a call is selected and the email

address of the calling person is known, the email placeholder will be filled with that email address. If the email address is not available, an empty string will be used.

- [LANG] This tag will be placeholder for the currently selected language in the Receptionist. The following placeholders are provided:
   en = English,
   fr = French,
   nl = Dutch,
   de = German,
   es = Spanish,
   da = Danish
   [THEME] (Only for the Webpage widget) This placeholder will be replaced with the name of the current theme, e.g. standard, dark, high contrast.
- [DESTINATION] (Only for the Webpage widget) This tag provides information about the called destination.
- [CONTEXT] (Only for the Webpage widget) This tag provides information about the caller and the user of the Receptionist.

## 9.5 Overview of all widgets

#### 9.5.1 Calendar widget

This widget shows the calendar information for a contact, when the calendar has been configured by your administrator. This widget is only available in the Premium edition.

The days of the week, that have to be displayed in the calendar in the week view mode, can be chosen in the configuration of the widget.



#### 9.5.2 Call flow widget

This widget shows all the destinations of the current incoming call, as well as the call history of that caller.

Call	flow			Œ
	Date and time	Reason	Destination	Duration
^	14-01-2021 15:14	Direct	Service Desk keu (tel:+31850068778)	0:29
		Direct	Service Desk changed queue-changed (+31850068778)	
(→	14-01-2021 15:14		Monica Schulz (55111)	
<b>(</b> >	14-01-2021 15:13		Monica Schulz (55111)	
(+	14-01-2021 15:11		Operator PeterConnects (0854863347)	
(+	14-01-2021 12:10		Operator PeterConnects (0854863347)	
<b>(</b> +	14-01-2021 11:06		Operator PeterConnects (0854863347)	
<b>*</b> +	14-01-2021 11:06		Operator PeterConnects (0854863347)	
¢+	14-01-2021 11:05		Operator PeterConnects (0854863347)	
<b>*</b> +	14-01-2021 11:05		Operator PeterConnects (0854863347)	
t+	14-01-2021 11:04		Operator PeterConnects (0854863347)	

It also shows previous calls up to a certain number of calls.

## 9.5.3 Contact information widget

This widget shows the basic contact information for the selected contact. Note that for contacts from some platforms, "extension" is displayed as "primary phone number" in the widget and "number" as "alternative phone number". When there is no extension, the number is displayed as primary phone number in the widget.

In the configuration it can be indicated, whether the contact notes should be displayed in this widget or not.

Settings for widget 'Contact information'		
Title (leave empty to use default title 'Contact information')		
Icon Select		
Contact information options           Image: Contact information options           Image: Contact information options		

#### 9.5.4 Contact note widget

With this widget, you can read and edit contact notes. Be aware that the contact note can be seen by anyone who has access to the same contact. This widget is only available in the Premium edition.

#### 9.5.5 Homepage widget

This widget is shown on the homepage screen, when the Receptionist is just started up.

#### 9.5.6 My call history widget

This widget shows the calls that you have received, including missed calls.

### 9.5.7 Twitter widget

This widget shows the timeline of a Twitter user. The Twitter widget can be added to the homepage. In the configuration of the widget, the Twitter screen name has to be entered. This can be only the screen name, the screen name preceded by '@' or the Twitter URL. These will all be converted into the screen name.

#### 9.5.8 Web Page widget

By inserting a Web Page widget, you can embed an external website into the Receptionist. For example, a page of a corporate website. This feature is only available in the Premium edition.

When inserting the Web Page widget, you can enter a URL of the page to embed:

Title (leave empty to use default title 'Web page')	
lcon	
e select	
Frame options	
Hide border	
Hide scrollbar	
URL*	
https://www.your-website.com/get-contact-data?email=[EMAIL]⟨=[LANG]]	
https://www.your-website.com/get-contact-data?email=[EMAIL]⟨=[LANG]]  A URL may contain the following placeholders:   IcLL: Will be replaced by the phonenumber of the contact. If not available, it will be replaced by an empty s  IcLANG: Will be replaced by the selected language in the Attendant.	tring. ty string.

In the frame options, it can be configured whether the border of the widget and/or the scrollbar of the page should be visible.

## Technical notes:

- The URL is mandatory. When you enter a url without a prefix ( 'http://', 'https://', or 'ftp://'), the prefix 'https://' will be added automatically when you save the configuration.
- The URL can contain the placeholders [EMAIL] and [LANG]. The placeholders are described in paragraph "URL placeholders for widgets" at the end of this chapter.
- Not all websites can be run in the web page widget, which is actually a <iframe> element. Not all websites allow this. Some websites have an X-frame-options set to any of the following values:
  - **deny**: The website is not allowed to run in an <iframe>.
  - **sameorigin:** The website can only be run in an <iframe> when they both have the same origin (domain)
  - **allow-from <uri>:** The website is only allowed to be run in an <iframe> by a website that is run from a specific uri.

The web page that is shown in the widget can be opened in a new window, using the  $\square$  button

## 1.1.1. Auto refresh of call history on homepage

The call history widget refreshes automatically. Newly handled calls are automatically added to the call history widget after hanging up or after a 30 seconds auto refresh period.



## 10 Help function

Via the main menu you can open the Help popup:

Soperator64@peterconn.routit.voipit.nl (64101)	🕙 Available	□ ↔
Preferences		ß
<b>i</b> Help		Q

## 10.1 Receptionist version

In the title of the Help popup, the current version of the Receptionist is mentioned.

Help (Broadsoft	1.3.3.7)	
📟 Shortcuts	📃 User manual	

## 10.2 Keyboard shortcuts

The first tab page of the Help menu describes the keyboard shortcuts, as described in the chapter about Keyboard navigation.

### 10.3 User Manual

On tab page 'User Manual' of the Help menu you can download the user manual.

Help (Broadsoft 1.3.3.7)	×
Shortcuts	
Download the user manual here	
	Close

#### 10.4 Application messages

On the tab 'Application messages' all toast messages are listed. The notifications can be copied with the 'Copy' button.

📟 Shortcuts	📒 User manual	Papplication messages	
07-05-2020 12:01:13 Email sent			→ Ģ
07-05-2020 12:00:57 Your changes have	been saved.		Ċ

#### 10.5 What's new

On the tab 'What's new', new features of the application are mentioned, belonging to the version of the application you are using. When you open the application with that version for the first time, the 'What's new' tab will automatically be shown.



## 11 Sign out

When you open the main menu (1), there is a menu item to sign out (2). If you click this item, you will be logged out of the Receptionist. You will not be asked for a confirmation.

💄 ope	erator65@bw.dev.onpeterconnects.com (0850068772)	1 🔮 Available 🗖
<b>\$</b> °	Preferences	0+ ▼ :
۵	Admin portal	
i	Help	doress or department
•	Sign out 2	

# 12 Global configuration

If you are permitted to edit the global configuration, you can go to the Admin Portal via the main menu:

💄 оре	erator65@bw.dev.onpeterconnects.com (0850068772)	🥝 Available	
¢°	Preferences	0+ 🔻	:
	Admin portal	ă <sup>r</sup> I	·
 i	Help	ddress or department	
•	Sign out		

In the Admin portal, you can enter some central configuration for the application. This configuration applies to all users of the Receptionist. The following paragraphs describe the configuration, that can be made.

## 12.1 Calendar

The Receptionist contains a calendar widget, that displays the calendar of the selected contact. This widget needs some central configuration before it can fetch the data of the calendar. That configuration is done in the Admin portal on the Calendar page.

note: this configuration is not available for the Receptionist that uses the Microsoft login, see the section describing the login options.

The screen shows a selection of which kind of calendar connector you want to use. 'None' is selected by default. The other options are 'Microsoft Office 365' and 'Google Suite'. These are described in the next paragraphs.

🛗 Calendar 🖂 Mail	×
Office connector	
Office connector for calendar information	
<ul> <li>None</li> <li>Microsoft office 365</li> <li>Google suite</li> </ul>	
None	
No calendar configuration	Save

## 12.1.1 Microsoft Office 365

苗 Calendar 🛛 🗠	Mail	×
Office conne	ctor	
Office connector	for calendar information	
None		
Microsoft office	ce 365	
Google suite		
Microsoft off	ice 365	
Tenant Id 🛛	How to find your Tepant Id?	
Client Id 🞱		
	Create application registration to access calendar information	
Client Secret 🧐	Create secret for application registration	
Required API Permission	Calendar.Read Assign API Permissions	
	Test	e

In this screen you can setup a connector for Microsoft Office 365.

To be able to read the calendars of your users, there needs to be an Application Registration in your Azure Tenant. After creating this App, you can fill in the Tenant Id, Client Id and Client Secret.

The field Required API Permission shows the permission you should assign to this app. **Be sure to assign** *application permissions* (and not *delegated permissions*).

For more information about these fields, you can click on the blue links below the fields. They will open a website in a new tab, containing information on how to configure those fields. These pages are publicly available websites that are not owned by the maker or the supplier of the Receptionist. Therefore, we cannot guarantee that these links will always be available.

When you have entered all data, you can click on the button "Test" to check if the settings you entered are correct. Then you can press the button "Save" to store those settings.

#### 12.1.2 Google Suite

🚔 Calendar 🛛 Mail				×						
Calendar connector										
Calendar connector for calendar inf	Calendar connector for calendar information									
None										
Microsoft Office 365										
🖲 Google Suite										
Google Suite										
Service Account ID	C Generate ID									
Api scope	https://www.googleapis.com/auth/calendar									
Google Suite	Managing API client access help page									
Test username 🛛										
				Test Save						

In this screen you can setup a connector for Google Suite.

First you have to generate the Service Account ID by a click on the "Generate ID" button. This generates your unique "Service Account ID", which you have to enter in your Google Suite.

The field "Api scope" has already been filled in for you. This is always the same.

The field "Google suite" contains a link that will open a webpage that contains information about connecting to the Google Suite from other applications. It will be opened in a new tab.

In order to test the configuration, you have to fill in a "Test username", which has to be a valid email address from your Google Suite. When you click on the button "Test", the configuration is tested. Then you can press the button "Save" to store those settings.

**Note:** If you configured the Google Calendar settings in the Admin portal, but the user still does not see the calendar events in the calendar widget, please check if the user has a valid email address that has enough rights to read the calendar of the selected contact.

#### 12.2 Mail

In the Receptionist, the user can send a message to one of the contacts, for example to leave messages that there was a call for that contact. This functionality works with a default setting.

You can add 1 or more additional destinations to which the e-mail is being sent as a blind copy (Bcc). This can be done on the tab Mail of the Admin portal.

Additional Destination configuration							
BCC address 🛛	test1@test.com; test2@test.com						

#### Add CC address on email settings

In addition to the existing blind copy (Bcc) feature in the email settings, users now have the capability to designate additional recipients for email copies directly through the Admin Portal's mail configuration. This enhancement allows for a more versatile distribution of correspondence.

#### **Key Features:**

Multiple Recipients: Users can specify one or more email addresses to receive a copy of the emails sent to contacts.

Easy Separation: Separate multiple email addresses using a semicolon (;) to ensure clear delineation of recipients.

Automatic Copying: Any email dispatched to a contact will concurrently be sent as a copy to the specified addresses in the Admin Portal's mail configurations.

Mail Mail Templates		₩ Connectors	🜲 Call Queue	User Privileges			
Addi	tional Destination	configuration					
BCC a	address 🛛	mahmoud	l.ali@anywhere36	5.net			
CC address 😢		pieterfran	pieterfrancois@gmail.com				

You can customize the e-mail configuration to work with your SMTP server by configuring this in the Admin portal. This can also be done on the tab Mail.

#### 12.2.1 SMTP Authentication

Additional Destination	configuration	
BCC address 0		
Dutgoing mail server co	onfiguration	
	G Use custom SMTP server to send mails •	
SMTP server •	smtp.office365.com	
Server part 🖤	587	
Default from address 🖲	stephen postma@pcadev.net	
	<ul> <li>Use default from address to send email</li> <li>Use user's email address to send email</li> </ul>	
	C Authentication required	
	• SMTP	
	Microsoft Office 365 The authentication user must be allowed to send email as the email addresses of the receptionist users, and as the default from address.	
Usemame	stephen.postma@pcadex.net	
Password	20002	0
fest Configuration		
Send testmail to •	peterpan@pcadev.net Test	

The following settings can be made:

- Use custom SMTP server to send mails: When this checkbox is checked, the configured settings will be used for sending mail.
- **SMTP server:** The name of the SMTP server that should send the mail.
- **Server port:** The port number of the SMTP server.
- **Default from address:** The email address that will be used for sending the emails, if the email address of the Receptionist user has not been configured.
- **Authentication required:** If the SMTP server requires authentication, you can check this checkbox to enable the fields where you can enter the credentials.
- **SMTP:** When selected, the authentication type SMTP will be used for sending mail.
- **Username:** The username that will be used for sending mail via the configured SMTP server.

- **Password:** The password that will be used for sending mail via the configured SMTP server.
- Send test mail to: Enter a valid email address to verify the configuration.

You can test your settings by clicking on the button "Test" and save your changes using the button "Save".

After setting the custom SMTP, the users need to login into the Receptionist again in order to see the new settings.

In case you also configured additional destination(s) these will also be used when sending emails using the custom SMTP settings.

Mail S Mail Templates	Connectors	A Coll Queue	User Privileges	Audio Files	Number Normalization	Contact Sources	C Ringback Tone Configuration	
Additional Destination co	nfiguration							*
BCC address								
CC address								
Outgoing mail server co	figuration							
SMTP server	O Use cu	stom SMTP serv	er to send mails 🔍					
Default from address	D lies de	fault from address	se to send email					
	<ul> <li>Use ut</li> </ul>	ser's email addre	ss to send email •					
	SMT	P posoft Office 365						

#### 12.2.2 Microsoft Office 365 Authentication

The following settings can be made:

- Use custom SMTP server to send mails: When this checkbox is checked, the configured settings will be used for sending mail.
- **SMTP server:** The name of the SMTP server that should send the mail.
- Server port: The port number of the SMTP server.
- **Default from address:** The email address that will be used for sending the emails, if the email address of the Receptionist user has not been configured.

- **Authentication required:** If the Mail server requires authentication, you can check this checkbox to enable the fields where you can enter the credentials.
- **Microsoft Office 365:** When selected, the authentication type Microsoft Office 365 will be used for sending mail.
- **Client Id:** The client Id that will be used for sending mail via the configured Mail server.
- **Client Secret:** The client secret that will be used for sending mail via the configured Mail server.
- **Tenant Id:** The tenant Id that will be used for sending mail via the configured Mail server.
- Send test mail to: Enter a valid email address to verify the configuration.

You can test your settings by clicking on the button "Test" and save your changes using the button "Save".

After setting the custom Mail server, the users need to login into the Receptionist again in order to see the new settings.

In case you also configured additional destination(s) these will also be used when sending emails using the custom Mail settings.

#### Important:

When using the Microsoft Office365 Authentication, following Permission are required for the related Enterprise application:

- Sign in and read user profile;
- Read all users' full profiles;
- Send mail as any user.

#### 12.2.3 Without Authentication

To be able to send mail without authentication, it first needs to create a connector in the tenant you want to use.

←	→ C admin.exchange.microsoft.com/#/connectors									
	Exchange admin center		♀ Search (Preview)							
≡		Home > Connectors								
ŵ	Home									
8	Recipients ^	Connectors								
	Mailboxes	Connectors help control the flow of email messages to and fro	om your Office 365 organization. We ector, since most organizations don't							
	Groups	recommend that you check to see if you should create a conne need to use them.								
	Resources									
	Contacts									
	Mail flow	+ Add a connector C Refresh								
	Message trace									
	Rules	Status ↑ Name	From To							
	Remote domains	On Send mail from pca-001-npd-we-cluster	r Your org O365							
	Accepted domains	On Send mail from pca-001-prd-we-cluste	Your org O365							
	Connectors									
	Alerts									
	Alert policies									

## Then set the settings in admin portal:

Mail Mail Templates	Vonnectors User Privileges Contact Sources
Additional Destinatio	n configuration
BCC address	
Outgoing mail server	configuration
	Use custom SMTP server to send mails
SMTP server	pcadev-net.mail.protection.outlook.com
Server port	25
Default from address 🔮	stephen.postma@pcadev.net
	O Use default from address to send email
	Use user's email address to send email
	Authentication required
	© SMTP
	O Microsoft Office 365
Test Configuration	
Send testmail to 9	peterpan@pcadev.net Test
	Save

The following settings can be made:

- Use custom SMTP server to send mails: When this checkbox is checked, the configured settings will be used for sending mail.
- **SMTP server:** The name of the SMTP server that should send the mail, use the specific DNS for the tenant.
- Server port: The port number of the SMTP server.
- **Default from address:** The email address that will be used for sending the emails, if the email address of the Receptionist user has not been configured.

#### Permissions

Applications can be granted permissions to your organization and its data by three methods: an admin consents to the application for all users, a user grants consent to the application, or an admin integrating an application and enabling self-service access or assigning users directly to the application. Learn more.

To request additional permissions for this application, use the application registration.

As an administrator you can grant consent on behalf of all users in this tenant, ensuring that end users will not be required to consent when using the application. Click the button below to grant admin consent.

	Grant	adm	in consent for PC	ADev							
Admin consent	User cons	ent									
	ons										
API Name	Ŷ	Ļ	Claim value	$\uparrow_{\downarrow}$	Permission	$\uparrow_{\downarrow}$	Туре	$\uparrow_{\downarrow}$	Granted through $\uparrow_{\downarrow}$	Granted by	$\uparrow_{\downarrow}$
Microsoft Graph											
Microsoft Graph	h		User.Read		Sign in and read user profile	2	Delegated		Admin consent	An administrator	
Microsoft Graph	h		User.Read.All		Read all users' full profiles		Application		Admin consent	An administrator	
Microsoft Graph	h		Mail.Send		Send mail as any user		Application		Admin consent	An administrator	

#### 12.3 User privileges via Admin portal

On tab page 'User Privileges' you can configure for all users the access to

- Admin Portal
- Add/edit/delete Personal contacts
- Add/edit/delete Shared contacts.
- Manage the Preferences
- Add/edit/remove Widgets

You can also configure the default settings for new users.

Note: the User Privileges tab is only available when logging in into the Receptionist Admin Portal with a Microsoft account.

#### 12.3.1 Setting user privileges for default user

With the '(Default for new users)', it is possible set the default for new users. For example: When the 'Enable Admin Portal access' and Enable manage widgets is unchecked, all new users do not get Admin Portal and manage widget access, they only get manage preferences, manage personal contacts and manage shared contacts.

Mail	Mail Templates	₩ Connectors	Call Queue	User Privileges	Audio Files
Man	age users privileg	es			
Sea	arch			٩	
	(Defaul	t for new users)		î l	
	Al	ain Deluxe	j		
	Alain	van den Berg			
	And	lrzej Deluxe			
	Arje	n Holleman			
	Bas van I	Leeuwen PcaDev			
	Brar	n the Prince	)	•	
User	rs privileges				
Defa	ult settings for new use	ers			
🗆 Ena	able Admin Portal acce	ess Ø			
🖾 Ena	able Manage Preferenc	ces Ø			
🖾 Ena	able Manage Personal	Contacts @			
🖾 Ena	able Manage Shared C	ontacts Ø			
🖬 Ena	able Manage Widgets	0			

#### 12.3.2 Setting user privileges for specific users

On tab page 'User Privileges' you can also configure the privileges for individual users.

When searching for a specific user, the privileges for this individual user can be set specifically for this user.
patar	
peter	٩
(Default for new users)	*
Peter Pan	
	w
Jsers privileges	*
Jsers privileges Settings for user Peter Pan	
Jsers privileges Settings for user Peter Pan 2 Enable Admin Portal access <b>2</b>	*
Jsers privileges Settings for user Peter Pan 2 Enable Admin Portal access 2 Enable Manage Preferences 3	*
Jsers privileges Settings for user Peter Pan 2 Enable Admin Portal access 2 Enable Manage Preferences 3 Enable Manage Personal Contacts 3	•

Remark: A user cannot remove the Admin Portal access for his/her own account. This is shown with a greyed-out checkbox.



Remark: When manage Personal and/or Shared contacts is disabled for a user, the selected user cannot manage (create, edit or delete) Personal and/or Shared contacts. The user can, however, still view Personal and/or Shared contacts and use them as a filter.

Favorite contacts (F6)	
	Personal contacts
Q Search for contacts	Shared contacts
	Corporate contacts
Peter Pan Peter Pan Peter Pan Peter Pan Peter Pan Peter Pan Peter Pan Peter Pan Peter Pan Peter Pan	··· 💌 🕻 🗖 🚑 External contacts
	Queues

# 12.3.3 Manage Profile Feature: User Manual

The Manage Profile feature is a powerful new addition to our platform that allows administrators to create, edit, and assign standardized settings profiles to multiple users. This feature enhances efficiency and consistency across the platform by enabling shared settings configurations while still allowing for user personalization.

#### 2. Key Functionalities

#### 2.1 For Administrators

- Create new profiles
- Edit existing profiles
- Delete profiles
- Assign profiles to users
- Unassign profiles from users

### 2.2 For Users

- Load assigned profile settings
- Utilize standardized settings across multiple users
- Maintain personal settings alongside profile settings

### 3. Accessing Manage Profile

Administrators can access the Manage Profile feature by clicking on "Manage Profiles" in the main menu of the Receptionist platform.

### 4. Creating a Profile

To create a new profile:

- 1. Navigate to the Manage Profiles menu
- 2. Click on "Create New Profile"
- 3. Enter a name for the profile

- 4. The current settings of your Receptionist instance will be saved as the default for this profile
- 5. Click "Save" to create the profile

### 5. Editing a Profile

To edit an existing profile:

- 1. Select the profile you wish to edit from the Manage Profiles menu
- 2. Click "Load Profile Settings" to apply the profile settings to your current instance
- 3. Make the desired changes to the settings
- 4. Click "Save Settings" to update the profile

#### 6. Deleting a Profile

To delete a profile:

- 1. Select the profile you wish to delete from the Manage Profiles menu
- 2. Click "Delete" option
- 3. Confirm the deletion when prompted

### 7. Assigning Profiles to Users

To assign a profile to users:

- 1. Access the Admin Portal
- 2. Navigate to the User Preferences tab
- 3. Select the user(s) you wish to assign a profile to
- 4. Choose the desired profile from the dropdown menu
- 5. Click "Save" to apply the changes

#### 8. Unassigning Profiles

To unassign a profile from a user:

- 1. Follow steps 1-3 from Section 7
- 2. Select "No Profile" from the dropdown menu
- 3. Click "Save" to apply the changes

### 9. Profile Merging

When a profile is applied to a user's account, the system performs a merge between the profile settings and the user's current settings. This merge functionality allows for greater flexibility and personalization.

### How Merging Works:

- Existing user settings are retained when they don't conflict with profile settings.
- Profile settings are added to the user's current configuration.
- In cases where both the user and the profile have settings for the same feature (e.g., widgets), both sets are combined.

### Example:

If a user currently has a Calendar widget, and the applied profile includes a Website widget, the user will see both widgets after the profile is applied. This merging feature ensures that users can maintain their personal preferences while benefiting from standardized settings provided by administrators.

# **10. Current Profile Settings**

As of now, profiles include the following settings:

- Joined queues
- Monitor queues
- Widgets

Please note that additional settings may be added in future updates based on customer needs and feedback.

### 11. User Experience

When a user logs into the Receptionist platform, they will see their personalized settings, which are a combination of:

- Their individually saved settings
- The settings from their assigned profile (if any)

These settings will persist across login sessions, ensuring a consistent user experience.

# 12. Troubleshooting

If you encounter any issues while using the Manage Profile feature, please try the following:

- Ensure you have the necessary permissions to create or edit profiles
- Log out and log back in to refresh your session
- Clear your browser cache and cookies

If problems persist, please contact our support team for assistance.

### 13. FAQs

Q: Can a user have multiple profiles?

A: No, a user can only have one profile assigned at a time.Q: What happens to a user's settings if their assigned profile is deleted?

A: The user will retain their current settings but will no longer receive updates from the deleted profile.Q: Can regular users create or edit profiles?

A: No, only administrators have the ability to create, edit, or delete profiles.For any additional questions or concerns, please contact our customer support team.

# 12.3.4

# 12.4 Maintain callback mail template

On tab page 'Mail Templates' you can configure mail templates used for call back e-mails. On the top half of the page, you see a list of mail templates. On the bottom half, you see the text to select a mail template or to create a new one.

Callback reminder (English)	+
Callback reminder (Danish)	↑ ↓
Callback reminder (Dutch)	
Callback reminder (French)	
Callback reminder (German)	
Callback reminder (Spanish)	

# 12.4.1 Adding a new mail template

In the top half of the page click on the sign to open the bottom part for entering a new mail template.

Mail 🛛	🖂 Mail Templates	₩ Connectors	🜲 Call Queue	User Privileges	🞜 Audio Files		x	Â
Mo	anage mail templat	es						l
	Callback remind	ler (English)	+					I
	Callback remin	der (Dutch)	$\uparrow$	$\downarrow$				I
	Callback remind	er (German)						I
	Callback remind	ler (French)						I
	Stepho	en						I
	Callback remind	er (Spanish)						
Mo	ail template							l
1	emplate name				Save			I
s	Subject							I
N	Nessage							I
								I
								I
								I
		Insert -			li			
						Cancel		
						Cancer		
								*

A mail template requires:

- 1. Template name, name of the template as it is shown to the user;
- 2. Subject, this line becomes the subject of the e-mail that is being sent;
- 3. Message: this contains the actual e-mail content. By using the insert drop down at the bottom the following fields can be inserted:
  - Name of caller;
  - Phone number of caller;
  - Date;
  - My mail address (e-mail address of the logged in user);
  - My name (name of logged in user);
  - My phonenumber.

Once the mail template is finished it can be stored by clicking the 'Save' button.

In case the mail template can be discarded, this can be done by clicking the 'Cancel' button.

#### 12.4.2 Arranging the order of mail templates

By selecting a mail template and clicking on the arrow up or arrow down button you can arrange the order of the mail templates. The mail templates are shown in this order when the user selects a mail template for sending a callback reminder e-mail.

Manage mail templates			
Callback re	minder (English)	+	
Callback re	minder (Danish)		
Callback n	eminder (Dutch)		
Callback re	minder (French)		
Callback re	minder (German)		
Callback re	minder (Spanish)		
Template name	Callback reminder (	English)	
Subject	Please call back: [C	NAMEJ [C_NUMBER]	
Message	Missed call from: [C Subject: Kind regards, [MY_NAME] Emait [MY_EMAIL] Phone: [MY_PHONE		

### 12.4.3 Changing or deleting a mail template or creating a new one

By selecting a mail template, the content of that mail template is shown in the bottom half of the screen. The user can change the content like with creating a new mail template and then press the 'Save' button.

By clicking the 'Delete' buttom a mail template that is obsolete can be deleted. The delete action has to be confirmed.

	inder (English)	+		
Callback ren	ninder (Dutch)	+ +		
Caliback rem	nder (German)			
Callback rem	inder (French)			
Ste	phen			
Caliback remi	nder (Spanish)			
Template name	Callback remin	der (English)	Totlete Save	
Subject	Please call bac	C [C_NAME] [C_NUMBER]		
Message	Missed cell fro Subject Kied regards, (MY_NAME)	n (C, NAME) (C, NUMBER)		
	ETRAIL IMPLEM			

### 12.5 Contact sources

In the Admin Portal, under the tab 'Contact Sources' additional contact sources can be managed. Contacts from these sources will become part of the contact list (F6).

Mail 🛛	🗖 Mail Templates	<b>₩</b> Connectors	🜲 Call Queue	🛡 User Pri	ivileges	5	🞜 Audio Files	🎝 Contact Sources	
Мс	anage Contact Sou	rces							
	E	Excel contacts			+				
		Star Wars			1	¥			
		New source							

A new contact source can be added via a link to the source document, this can be Excel or CSV source documents.

Each contact source appears in the filter menu and can be toggled on/off just like other contact types.

After adding a contact source, an application refresh is necessary to make the contact source show up in the filter menu. The contact list (F6) will automatically be updated when the contacts in the source document are changed, however it might take some time for the new changes to end up there.



The order in which the contact sources are presented in the filter menu can be changed by clicking on an item in the list in the Admin Portal and using the arrow up and down buttons to the right to move it up or down the list, respectively.

Manage Contact Sources	
Excel contacts	+
Star Wars	↑ ↓
New source	$\smile$

### 12.5.1 Managing access

In order to work with contact sources, we need to set up a way for the attendant console to be able to read files from Sharepoint. At the top, there is a panel to manage access.

There are 3 possible ways to set up access:

1. Use the Anywhere365 enterprise application. For this, you only need to specify your own Azure tenant ID, save it and give consent. The Anywhere365 then has access to read the files your tenant's Sharepoint.

Manage Access		*
To use the Contact So	purce feature, please set up access to shared files on Sharepoint.	
Read files with Any	where365 enterprise application	
○ Read files with own	application registration 🔮	
<ul> <li>Read files with use</li> </ul>	r access 🔮	
	(	
Directory (tenant) ID	8939ca96-dcdc-4651-b5e0-71d6162eb5d0	
·	How to find your Tenant ID?	
Required API	Microsoft	
Permission	Graph/Files.Read.All	
	s	ave Give consent

2. Create an application registration in your own tenant and configure the access to it here. You will need to set up the permissions yourself. The Client Secret is stored securely and after saving will not be shown in the admin portal. The links below each setting will help gather the required information and to set up the application in Azure.

Manage Access		*
To use the Contact So	urce feature, please set up access to shared files on Sharepoint.	
<ul> <li>Read files with Anyw</li> </ul>	where365 enterprise application 🙆	
Read files with own	application registration	
○ Read files with user	access 🛛	
Directory (tenant) ID	8939ca96-dcdc-4651-b5e0-71d6162eb5d0	=
0	How to find your Tenant ID?	
Application (client) ID	6e63a1ef-38df-4491-988b-983604766c94	
0	Create application registration to access shared contact files on Sharepoint	
Client Secret		۲
	Create secret for application registration	
Required API	Microsoft	
Permission	Graph/Files.Read.All	
	Assign API Permissions	
		Save

3. Use delegated access from your user. This option is only available if you are logged in with an Azure User in the Attendant Console. This also requires admin consent. Since different users can be used to access the files, extra caution must be taken when setting up the access to the files in Sharepoint. If some users do not have access to the file, it might lead into issues when loading the contacts.

Manage Access		*
To use the Contact	Source feature, please set up access to shared files on Sharepoint.	
○ Read files with A	nywhere365 enterprise application 🔮	
○ Read files with o	wn application registration 🔮	
Read files with us	ser access 🔨	
Required API Permission	Microsoft Graph/Files.Read.All	
	Save	Give consent

The Manage Access panel will be collapsed when there are contact sources configured, but the settings can always be revisited by clicking on the header.

### 12.5.1.1 Admin consent

As said before, the application requires admin consent for reading files users have access to. If this consent is not given, a message will pop up including a button to give consent. If no admin consent has been given, validating a contact source will result in a permission denied error.

This feature requires consent for reading files accessible to the user. 🔮	Give consent	
Manage Contact Sources		
	+	

Clicking on the 'Give consent' button will redirect you to a Microsoft page where you can accept the permissions requested:



After giving consent, it might take a few minutes before it has been processed. Please refresh the page until the message disappears. Once the consent message is no longer presented, the contact sources are available to be configured.

### 12.5.2 Adding or changing a contact source

By clicking the plus icon interview in the list of current contact sources you can add a new contact source. After clicking the button, a new section opens below the list of contact sources. You can also select one of the contact sources in the list to edit.

Edit		
Name		
Url		
Icon	Select icon	
	Validate Save and import	
		Cancel

In the form you can give the source a name and icon which is displayed to the users in the contact list filter menu. You can also paste the URL of the file here.

Files can be either Excel files or Csv files stored on the organizations SharePoint or OneDrive.

Edit		
Name	Extra Contacts	
Url	https://company.sharepoint.com/:x:/g/Eb8ARR9qVbJlku2b3401PjsBE1Q6EVwQ	
lcon	🐣 Select icon	
	Validate Save and import	
		Cancel

To create a URL which can be pasted in this form, go to your organization's SharePoint or OneDrive in your browser and locate the file. Then click on the Share Link button:

٦

Γ

+ New $\sim$	$ar{\uparrow}$ Upload $\smallsetminus$	Edit in grid view	ಢ Sync 🛛 🛱 Add sho
Documents	5		
Ľ	Name $\vee$		Modified $\vee$
X	contacts.xlsx	È :	Monday at 1:32 AM

There you can copy the link and change access. By default, the Edit rights seem to be granted for everyone who has the link.

You have to change the settings by clicking on "Anyone with the link can edit".

Send link contacts.xlsx	×
$\bigoplus$ Anyone with the link can edit >	
To: Name, group or email	$\mathcal{O}$ $\sim$
Message	
6	Send
Copy link	
Anyone with the link can edit >	Сору
Shared with:	

Only view access is required for the contact source functionality: uncheck the 'Allow editing' checkbox to make it read-only.

Additionally, make sure to make the link only works for people in the same organization. The option "Anyone with the link" is unsecure, while "People with existing access" or "Specific people" might not allow the system to read the contacts when the document has been changed.

Link settings contacts.xlsx	×
Who would you like this link to work for? Learn mo	pre
People in PCADev with the link	~
People with existing access	
(R) Specific people	
Other settings Allow editing	
Block download	
Appl	y Cancel

A pop up is presented with a 'Copy' button to copy the URL to the contact source to the clip board. Please verify that it indeed indicates that only people from your tenant with the link can view the document.



With the URL on your clipboard, please return to the admin portal. After filling in both the Name and the URL and optionally selecting a different icon, you have to validate the contact source by clicking the 'Validate' button.

Edit	
Name	Extra Contacts
Url	https://company.sharepoint.com/:x:/g/Eb8ARR9qVbJIku2b3401PjsBE1Q6EVwQv
lcon	Select icon
	Validate Save and import
	Cancel

This will validate and test access to the contact source. After validation a message will appear to show the user whether the file is correct or not. In case the file is not valid a detailed message is shown, in case the file is valid it shows 'Validation is successful'.

Name       Extra Contacts         Url       https://company.sharepoint.com/:x:/g/Eb8ARR9qVbJlku2b3401PjsBE1Q6EVwQv         Icon       Select icon         Validate       Save and import         Validation is successful.       Validation is successful.	Edit		
Url       https://company.sharepoint.com/:x:/g/Eb8ARR9qVbJlku2b3401PjsBE1Q6EVwQv         Icon       Select icon         Validate       Save and import         Validation is successful.       Validate	Name	Extra Contacts	
Icon  Select icon Validate Save and import Validation is successful.	Url	https://company.sharepoint.com/:x:/g/Eb8ARR9qVbJlku2b3401PjsBE1Q6EVwQ	
Validate Save and import Validation is successful.	lcon	Select icon	
Validation is successful.		Validate Save and import	
		Validation is successful.	
Cancel			Cancel

After a successful validation, the 'Save and import' button will be enabled. Click it to actually import and save the new contact source in the contact list (F6).

Edit		
Name	Extra Contacts	
Url	https://company.sharepoint.com/:x:/g/Eb8ARR9qVbJIku2b3401PjsBE1Q6EVwQv	
lcon	Select icon	
	Validate Save and import	
	Validation is successful.	
		Cancel

A progress message will appear showing the status of the import.



Once ready it shows a green 'Ready'.

Edit		
Name	Extra Contacts	
Url	https://company.sharepoint.com/:x:/g/Eb8ARR9qVbJlku2b3401PjsBE1Q6EVwQv	
Icon	🛎 Select icon	
	✓ Ready C Refresh cache	
	Validate         Save and import         Telete	
		Cancel

The contacts in the new contact source will only show up in contact list search results when all contacts have been imported and actually saved (available in the cache). After login the Attendant will see the new contact source in the contact list filter menu.



Changes to the underlying contact source files will automatically be picked up when a search is being performed and the file has been found to be modified. Important to know is: it might take some time before the changes are synchronized into the contact list, especially with large contact source files (it triggers a cache rebuild).

When the cache is ready, a 'Refresh cache' button appears. This button will force a complete refresh of the contacts in the source file.

Cache status:	✓ Ready	CRefresh cache

Note: after changing a document in Sharepoint or OneDrive, it may take a couple of minutes before the changes are picked up by the Admin Portal.

In case you do not want to create the contact source after all, you can click 'Cancel' and the bottom part is closed and nothing is saved.

Mail Mail Templates	Connec msteams3.dev.onpeterconnects.com says Are you sure you want to lose changes tacts trce
Extra Cor	tacts
Edit Name	Not needed
Url	https://company.sharepoint.com/:x:/g/EXLD54MbEMJFge6ef73qEeAB4z6e-Xd1
Icon	Select icon     Save and Import
	Cancel

#### 12.5.3 Deleting a contact source

To delete a contact source, click on it in the list of contact sources. The form with the details will appear. Click the 'Delete' button confirm that you are sure you want to delete this contact source.

Extra Contacts	
Edit Name Extra Contacts	
Url https://company.sharepoint.com/:x:/g/EXLD54MbEMJFge6ef73qEeAB4z6e-Xd\	
Icon Select icon Cache status: ✓ Ready Z Refresh cache Validate Save and Import Telete	

In the background, the contacts will be removed. The contact source will also disappear from the filter menu in the contact list (F6).

### 12.5.4 Supported file formats and columns

Currently the following files are supported: Excel files (.xlsx, .xls), comma or semicolon separated files (.csv) and tab separated files (.tsv).

For Excel files the first sheet must contain the data, all other sheets are ignored.

On the first row the column names are specified. Every column corresponds to a property of a contact. See below for a list of all properties supported. Required properties are the LastName and Id.

The next rows should contain the information of the contacts.

For template and example files, please see the following zip file: <u>https://delivery.anywhere365.io/deployment/Attendant-Receptionist-Contact-Sources.zip</u>

Below is a list of supported columns:

Column	Description
Id	(Required) A unique ID for this contact in this
	file.
FirstName	
LastName	(Required)
DisplayName	
CompPrimaryPhone	Primary phone number
CompAltPhone	Alternative phone number
MobilePhone	
Pager	*)

Titles	*)
CompanyEmail	
Building	**)
BuildingAddress	*)
BuildingPostalCode	*)
BuildingCity	*)
BuildingCountry	*)
Location	*)
Department	*)

\*) field not displayed when using login via Webex.

\*\*) field not displayed when using Broadworks login.

### **1.1.1. External Contact Sources**

It is possible to configure additional contact sources such as Excel files with contacts in the Admin Portal. In the admin portal the files can be uploaded.

Admin Consent for Files.Read.All is required to be able to configure the sources.

To manage the access to the SharePoint location for external contact sources, there are 2 or 3 options:

- 1. Read files with Anywhere365 enterprise application
- 2. Read files with own application registration
- 3. Read files with user access (only Microsoft login)

My profile 🚔 Calendar	Mail Mail Templates	4 Contact Sources	Company directory		
Manage Access					*
To use the Contact Source t	eature, please set up access to shar	ed files on Sharepoint.			
Read files with Anywher	e365 enterprise application 🛛				
O Read files with own app	lication registration				
Directory (tenant) ID 🛛	c6f How to find your Tenant ID?				-
Required API Permission	Microsoft Graph/Files.Read.All				
				Save Give consen	t

After giving consent by an admin user, new contact sources can be uploaded in the Admin Portal. Existing contact sources can be maintained or deleted here as well.

💄 My profile 🛛 🛱 Ca	alendar 🛛 Mail	Mail Templates	4 Contact Sources	🗘 Company dire	story		
Manage A	ccess						×
Click here to set u	p access to read conta	ct source files.					
Manage C	ontact Source	s					
	first f	īle me	+	•			
Edit							
Name		first file				<u>ا</u>	
Url		https://petercor	nnectsdevops.share	point.com/:x:/s	/Testsite2/EVhLG	YD-BjpLr56RbMn	
Icon		\$ Sele	ct icon				
Cache status:		✓ Ready     ✓ Ready <b>२</b> Ready       Validate   Sa	ve and import	Delete			
							Cancel

L.

The receptionist users see the contacts form external contact sources in the F6 list and as one of the filter options:

onversations (F2)	
My calls (0)	3 My call history
<ul> <li>Q. Service Desk (0)</li> </ul>	
A Service Desk queue 2 (0)	
BasicCallCenter (0)	> C→ John2 Prestwood2
<ul> <li>Testcomplete Monitored (0)</li> </ul>	> 🚱 0851122230
© Queue2 (0)	
<ul> <li>Primary Queue 58 (0)</li> </ul>	D Personal contacts
	Corporate contacts
	Group contacts
=	D & Queues
Itered contacts (F6)	🗖 📚 first file
Search for contacts	🙄 i 🗖 揉 anyname
John Prestwood	(All contacts)
<b>译 法</b>	operatorfilter
George Price	O 28 All Monica
🚺 🔆 📩 🗖	O 🐉 all the john
Monica 53	O Igi Japio
	A set Marc Test (Operators only)
	O S Marc rest (operators only)
Operator PeterConnects	O S Matchest (Operators only)
Operator PeterConnects	O Pca
Operator PeterConnects	O Perator53     O Perator53     O Perator53     O Stephen (department testin
Operator PeterConnects	A matchest (operators only)     A matchest (operators onl

For further information regarding this functionality, see User Manual section Global Configuration/Contact Sources.

# 12.6 Enterprise Group selection

In the tab Company directory, the display of the contacts in the F6 (Contact list) can be set to:

1. Enterprise directory

2. Group directory

In case Enterprise is selected all contacts of the entire enterprise are shown in F6, in case Group is selected, only the contacts of the group that the user that is logged in belongs to are shown.

When no selection is made yet, by default all contacts of the enterprise are shown in F6

Ë	Calendar 🖂 Mail 🖂 Mail Templates 🚓 Company directory
	Company directory search preference
	Select the directory which will be used to display F6 contacts for this tenant.
	Enterprise directory
	○ Group directory
	Save

Note: after changing and saving a new setting, it can take up to 24 hours before the setting is reflected in the search results of the F6 contact list.

Some organizations do not support this functionality, in those cases no selection can be made, and the selection functionality is disabled.



# 13 Change history

1.15	Update §4.14 with callback numbers of previous calls.
1.14	Update section §4.11 with option to add/edit/remove contacts in F6
	<ul> <li>Added §4.12 about adding shared contacts when using Microsoft login</li> </ul>
	<ul> <li>Added §4.13 describing the use of additional contact sources.</li> </ul>
	<ul> <li>Updated §6.1 adding the ability to change the presence status.</li> </ul>
	<ul> <li>Added §12.3 about managing the user privileges when using Microsoft login</li> </ul>
	<ul> <li>Added §12.5 about managing contact sources</li> </ul>
1.13	Added info about required permissions for mail configuration via Office365
	authentication in §12.2
	<ul> <li>Added in §12.2 the option for configuring mail without authentication</li> </ul>
	Added §12.4 for selection between enterprise and group directory
1.12	• Chapter 2, added §2.2, §2.3, §2.4 related to the Microsoft/Webex login flow
	<ul> <li>Updated section 4.10.1.3 to list search limitations for webexbw login.</li> </ul>
	<ul> <li>Updated §4.8 with option to dial mobile number from F6</li> </ul>
	<ul> <li>Updated 4.10.4 related to advanced search</li> </ul>
	Updated image in chapter 8
	Added a note to §12.1
	Updated §12.2 Mail configuration now has test email address field.
1.11	<ul> <li>Added §12.3 in which maintaining the mail templates is described</li> </ul>
	<ul> <li>Updated §4.12 in which the way the maintainable mail templates are being used is described</li> </ul>
1.10	<ul> <li>Added §4.12.1 about opening the email form in a new window</li> </ul>
	<ul> <li>Inserted §7.2.4 regarding calling timeout setting and changed picture in §7.2</li> </ul>
1.9	• Reorganized § 4 about the contact list. Searching and filtering are now grouped
	under the same paragraph
	• Updated §4.11 about editing contacts, to reflect the new way of deleting a
	personal contact.
	<ul> <li>Updated §4.13 about contact notes, to reflect the new way of deleting a</li> </ul>
	contact note.
	<ul> <li>Added § 4.10.3 about saving search queries</li> </ul>
	<ul> <li>Added §4.10.4 about advanced search</li> </ul>
	<ul> <li>Added § 5.9.2 about transferring to other contact numbers</li> </ul>
	<ul> <li>Updated and added text and a new image in § 12.2, you can configure bcc e- mail addresses</li> </ul>
1.8	• Updated image in § 7.3 and text and images related to ACD in § 7.3.1.1

1.7	<ul> <li>Added new text and images about sorting favorites in § 4.10.3</li> </ul>
	• Added text and images in § 7.3.1.1 that the current ACD state is shown in the
	main menu
	<ul> <li>Added new shortcuts to key navigation in § 8</li> </ul>
	• Rewritten § 9.1. Zooming the widget is now maximizing/restoring the widget
	• Changed the screenshot in § 9.2 and added a note about the content of the
	widgets not being visible in edit mode.
	• Added a note in § 9.5.8 about automatically adding https:// to the url of the
	web page widget
1.6	• Chapter 4 has been rewritten to match all new features in the contact list.
1.6	<ul> <li>Chapter 4 has been rewritten to match all new features in the contact list.</li> <li>Added § 5.1 about the connection to the server</li> </ul>
1.6	<ul> <li>Chapter 4 has been rewritten to match all new features in the contact list.</li> <li>Added § 5.1 about the connection to the server</li> <li>Added 'Handled by' in § 5.2</li> </ul>
1.6	<ul> <li>Chapter 4 has been rewritten to match all new features in the contact list.</li> <li>Added § 5.1 about the connection to the server</li> <li>Added 'Handled by' in § 5.2</li> <li>Added § 5.9 about selecting a device to start and answer calls</li> </ul>
1.6	<ul> <li>Chapter 4 has been rewritten to match all new features in the contact list.</li> <li>Added § 5.1 about the connection to the server</li> <li>Added 'Handled by' in § 5.2</li> <li>Added § 5.9 about selecting a device to start and answer calls</li> <li>Updated the image for receptionist status 'Do Not Disturb' in § 6.2</li> </ul>
1.6	<ul> <li>Chapter 4 has been rewritten to match all new features in the contact list.</li> <li>Added § 5.1 about the connection to the server</li> <li>Added 'Handled by' in § 5.2</li> <li>Added § 5.9 about selecting a device to start and answer calls</li> <li>Updated the image for receptionist status 'Do Not Disturb' in § 6.2</li> <li>Added Date/Time format to § 7.1 (Interface preferences) and changed the</li> </ul>
1.6	<ul> <li>Chapter 4 has been rewritten to match all new features in the contact list.</li> <li>Added § 5.1 about the connection to the server</li> <li>Added 'Handled by' in § 5.2</li> <li>Added § 5.9 about selecting a device to start and answer calls</li> <li>Updated the image for receptionist status 'Do Not Disturb' in § 6.2</li> <li>Added Date/Time format to § 7.1 (Interface preferences) and changed the Preferences picture</li> </ul>
1.6	<ul> <li>Chapter 4 has been rewritten to match all new features in the contact list.</li> <li>Added § 5.1 about the connection to the server</li> <li>Added 'Handled by' in § 5.2</li> <li>Added § 5.9 about selecting a device to start and answer calls</li> <li>Updated the image for receptionist status 'Do Not Disturb' in § 6.2</li> <li>Added Date/Time format to § 7.1 (Interface preferences) and changed the Preferences picture</li> <li>Inserted § 9.5.2 about the call flow widget</li> </ul>

1.5	Added text to § 3.1 (Resizing the panels)
	Changed text in chapter 4 (The contact list)
	<ul> <li>Added text about dial pad in § 4.2 (Searching a contact)</li> </ul>
	• Changed the pictures of the presence/hook state in § 4.4 (hook states)
	<ul> <li>Changed text in § 4.6 (Filtering of contacts)</li> </ul>
	• Added a note to § 4.7.1 and § 4.7.2 that the image might show different fields
	depending on your edition
	<ul> <li>§ 4.9 of making a contact note has been rewritten</li> </ul>
	<ul> <li>Added § 5.1.4 (Greyed out pickup button)</li> </ul>
	• Changed the introduction of chapter 7, so it no longer sums up the tab pages
	that are in the preferences popup
	Changed title of § 7.1 from Layout preferences into Interface preferences
	<ul> <li>Added tabbed layout to § 7.1 (Interface preferences) and changed the</li> </ul>
	Preferences picture
	<ul> <li>Moved text from § 7.2.2 and § 7.2.3 to § 7.4 (Availability)</li> </ul>
	Changed title of § 7.4 from Profile into Availability
	<ul> <li>Changed text and pictures in § 9.3 (Adding a widget)</li> </ul>
	<ul> <li>Moved § 9.4.1 (detailed description of the Web page widget) to § 9.5.7</li> </ul>
	• Elevated § 9.4.2 (URL Placeholders for widget) one level higher to § 9.4 and
	added destination and context to this paragraph
	• Added sub paragraphs to § 9.5 (Overview of all widgets), so each widget has its
	own paragraph
	Added text and picture to § 9.5.1 (Calendar widget)
	• In § 9.5.2 of the contact information widget, the option to show or hide the
	contact notes was added
	Added § 9.5.6 (Twitter widget)
	Updated screenshot in § 9.5.7 (Web page widget) and added text about the     artise to show an hide the boundary and smallher
	option to show or hide the border and scrollbar
	• Changed the text in § 10.3 about the user manual
1.4	Added § 10.5 (What s new)
1.4	<ul> <li>Added picture of error code for failed login to § 2</li> </ul>
	<ul> <li>Changed § 4.1 due to drop-down menu for buttons</li> </ul>
	<ul> <li>Changed § 4.5 due to favorites button now also available in F6</li> </ul>
	<ul> <li>Added language, added theme Dark and replaced screenshot in § 7.1</li> </ul>
	Added § 7.2.5 about Audio Alerting
	• Updated the screenshot of shortcuts in chapter 8 about keyboard navigation
	• Updated screenshot in § 9.4.1 due to the new help text and new frame options
	in the Webpage widget
	Renamed startup widget to homepage widget in § 9.5
	Inserted § 10.4 about Application messages

1.3	
	Added when a specific feature is only available in a certain edition
	Changed description of search box in § 4.2
	<ul> <li>Added "Idle", "Call forwarding" and "Do not disturb" to § 4.4 (hook states)</li> </ul>
	• <b>Removed § 4.5</b> about Do Not Disturb. This state is integrated in § 4.4 now. This
	means § 4.6 to § 4.10 are now shifted up one paragraph
	Inserted § 7.3.1.1 (ACD state)
	• Replaced screenshot in §4.8 (send mail) and added text about field 'Send from'.
	<ul> <li>Changed the bold titles in § 5.1 into three subparagraphs:</li> </ul>
	§ 5.1.1 (call states),
	§ 5.1.2 (destination information,
	§ 5.1.3 (call actions).
	Each subparagraph has been given an introductory sentence.
	Added 'call forwarding' to the possible destination information in 5.1.2
	Rewritten chapter 6 and renamed it from 'Presence' to 'Receptionist status'.
	<ul> <li>Renamed §7.2.1 from 'Call handling settings' to 'Call handling'</li> </ul>
	• Inserted §7.2.2 (Do not disturb) and §7.2.3 (Forward all calls).
	This implies that the old §7.2.2 (busy camp on) now has number §7.2.4. Also,
	that paragraph is renamed from 'Enable or disable Busy camp on' to 'Busy
	camp on'.
	Chapter 11 about sign out has been changed
1.2	Inserted chapter 12 (Global configuration)
1.2	<ul> <li>Added 'Remember username' to § 2 Login page</li> </ul>
	<ul> <li>Updated an image and text in § 3 Receptionist main screen</li> </ul>
	<ul> <li>Replaced the queue icon in chapter 4 and image in § 4.1</li> </ul>
	<ul> <li>Added § 4.5 (Do not disturb)</li> </ul>
	<ul> <li>Removed former § 4.5.1 and § 4.5.2 and changed the text of § 4.6 Favorites</li> </ul>
	<ul> <li>Added § 4.7 (Filtering contacts) including § 4.7.1</li> </ul>
	<ul> <li>Changed text and pictures in the subparagraphs of § 4.8 Editing contacts</li> </ul>
	• Added the language component and changed the image to § 4.9 Sending mail
	Added § 4.10 Making a contact note
	Added destination, a call state and a call action to § 5.1
	<ul> <li>Added text regarding pressing F5 in § 5.5</li> </ul>
	-
1	<ul> <li>Inserted § 5.8 (Calling other numbers of a contact)</li> </ul>
	<ul> <li>Inserted § 5.8 (Calling other numbers of a contact)</li> <li>Replaced screenshot in § 7.2 (Conversation settings)</li> </ul>
	<ul> <li>Inserted § 5.8 (Calling other numbers of a contact)</li> <li>Replaced screenshot in § 7.2 (Conversation settings)</li> <li>Replaced screenshot in § 7.3 (Queue management)</li> </ul>
	<ul> <li>Inserted § 5.8 (Calling other numbers of a contact)</li> <li>Replaced screenshot in § 7.2 (Conversation settings)</li> <li>Replaced screenshot in § 7.3 (Queue management)</li> <li>Replaced screenshot in § 7.4 (Profile settings)</li> </ul>
	<ul> <li>Inserted § 5.8 (Calling other numbers of a contact)</li> <li>Replaced screenshot in § 7.2 (Conversation settings)</li> <li>Replaced screenshot in § 7.3 (Queue management)</li> <li>Replaced screenshot in § 7.4 (Profile settings)</li> <li>Changed text and image in § 9.2</li> </ul>
	<ul> <li>Inserted § 5.8 (Calling other numbers of a contact)</li> <li>Replaced screenshot in § 7.2 (Conversation settings)</li> <li>Replaced screenshot in § 7.3 (Queue management)</li> <li>Replaced screenshot in § 7.4 (Profile settings)</li> <li>Changed text and image in § 9.2</li> <li>Added [THEME] placeholder in § 9.4.2 (URL placeholders for widgets)</li> </ul>
	<ul> <li>Inserted § 5.8 (Calling other numbers of a contact)</li> <li>Replaced screenshot in § 7.2 (Conversation settings)</li> <li>Replaced screenshot in § 7.3 (Queue management)</li> <li>Replaced screenshot in § 7.4 (Profile settings)</li> <li>Changed text and image in § 9.2</li> <li>Added [THEME] placeholder in § 9.4.2 (URL placeholders for widgets)</li> <li>Changed contact information and contact notes widget text and added</li> </ul>
	<ul> <li>Inserted § 5.8 (Calling other numbers of a contact)</li> <li>Replaced screenshot in § 7.2 (Conversation settings)</li> <li>Replaced screenshot in § 7.3 (Queue management)</li> <li>Replaced screenshot in § 7.4 (Profile settings)</li> <li>Changed text and image in § 9.2</li> <li>Added [THEME] placeholder in § 9.4.2 (URL placeholders for widgets)</li> <li>Changed contact information and contact notes widget text and added calendar widget in § 9.5</li> </ul>

1.1	Replaced screenshot and added text about 'forgot password' in chapter 2.
	Replaced screenshot in § 5.1 (Call states and call actions). Also added the action
	"Decline an incoming call" in the overview of call actions at the red hang up button.
	Inserted § 5.7 Working with queue calls
	The old § 5.6 'Restriction in picking up queue calls' is moved to § 5.7.3
	Replaced screenshot in § 7.2 (Preferences - Conversation)
	Inserted new paragraph § 7.3 (Queue management) and moved the old § 7.2.1
	(Queue management) there. The text has been altered and the minimum queues
	has been set to 10 (used to be 5)
	Inserted paragraph § 5.5 Call recording options
1.0 R6	Replaced the last two images in § 4.6.1 (adding a contact)
	Replaced both images in § 4.6.2 (editing a contact)
	Replaced first image in § 4.6.3 (deleting a contact)
	Updated second image in § 4.7 (Send mail) and updated the text about canceling
	the mail.
	Replaced image in § 7.1 (Preferences - Layout)
	Replaced first image in § 7.2 (Preferences - Conversation)
	Replaced image in § 7.3 (Preferences - Profile)
	Fixed spelling "Company wide" $\rightarrow$ "Companywide"
1.0 R5	Inserted § 4.7 Sending a mail to a contact
1.0 R4	Updated screenshot of conversation settings in § 7.2
	Rewritten § 7.2.1 (Queue management)
	Rewritten and renamed § 7.2.2 (inserted Auto Answer)
	Removed option 3 from § 5.3 ("Drag and drop" a call to another call (announced
	transfer)) and changed text and picture in 5.3 point 1.
	Removed drag & drop text from § 5.5
	Changed formulation of 'directed call pick up' in § 5,2
1.0 R3	Added contact type / favorite icons to images:
	- intros of chapter 3 and 4
	- paragraphs 3.1, 4.1, 4.3, 4.5, 9.1, 9.2
	Updated § 4.4 about presence and hook state and their tool tips
	Inserted § 5.2 Directed call pickup
	Updated screenshot of online help in chapter 8
	Inserted § 5.5 Conference calls
	Changed "connected call" icon 🔍 from green to red in all images where applicable
	(chapter 3, § 3.1, § 5.3, § 5.4, § 5.5)
	Renamed "press" to "click on" whenever it involves a button.
	Added a new way of adding a contact in § 4.6.1
	Updated screenshots and description of conference calls § 5.5
	Inserted § 4.2.1 restriction of search results depending on license
	Inserted § 5.6 restriction of picking up queue calls
	Updated § 7.2 (rewritten queue management) and added 3 sub-paragraphs.
	Removed explanation about setting presence from chapter 6

1.0 R2	Applied chapter and paragraph numbering
	Added change history
	Updated screenshots and some text in § 4.6 (Editing contacts)
	Temporarily removed § 5.4 (Conference calls)
	Added chapter 6 about setting your presence
	Updated § 7.2 (Conversation settings)
	Added § 7.3 (Profile settings)
	Updated § 9.4.1 (Web page widget)
	Temporarily removed § 9.4.2 (Chart widget)
	Temporarily removed § 9.4.3 (Conversation history widget)
	Updated § 9.5 (Overview of all widgets)
1.0	Initial version